COLLECTIVE AGREEMENT

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OF ST. ALBERT

AND

THE UNITED STEELWORKERS OF AMERICA LOCAL 5885

Effective: April 1,2000 to March 31, 2003

Youville.ca.2003.doc

Table of Contents

Page

Article 1.	Term of Agreement1
Article 2	Definitions1
Article 3	Union Recognition.
Article 4	Union Membership and Dues Deduction
Article 5.	Humanity Fund
Article 6.	Bulletin Board Space,
Article 7.	Management Rights
Article 8.	No Discrimination
Article 9.	Probationary Period
Article 10.	Seniority
Article [].	Appointments and Transfers
Article 12	Annual Performance Appraisals
Article 13	Hours of Wark
Article 14.	Shiftsand Schedules
11000 1 1	14.01 Shift Operation 10
	14.02 Schedulingof Shifts, 10
	14.03 Temporary Shift Pattern Reassignment
	14.04 Schedule Posting and Schedule Changes
	14.05 ShiftExchanges
	14.06 Reporting Pay
	14.07 Copies of Schedule
Article 15.	Overtime
Article 16.	Salaries
	16.01 Application of Salaries Schedule
	16.02 Change in JobContent
	16.03 Creation of a New Classification
	16.04 Attendance at Meetings and Special Events
Article 17.	No Pyramiding
Article 18.	Shift Differential and Weekend Premium
Article 19.	Transportation
Article 20.	Named Holidays
Article 21.	Annual Vacation 17
	21.01 Definition
	21.02 Vacation Entitlements 18
	21.03 Cessation of Vacation Accrual
	21.04 Time of Vacation
Article 22.	Employee Benefits
Article 23	pension Plan21

Page	
21	

Article 24.	SickI	001/0		21
Atticle 24	34.01	Annli	ation of Sick Leave	01
			all of Sick Leave Credits,	
			ion of Sick Leave Credit Accrual	
	24.04	niness	Reporting	44
			eave Pay Entitlement	
			al and Dental Appointments	
			st For Sick Credit Balance	
			When Sick Bank Exhausted	
Article 25.			npensation	
Article 26				
Article 27.			sence	
			al Leaves	
			nity Leave	
	27.03		tion Leave	
			Appearance	
			vement Leave	
			al Leave	
Article 28.			cation	
Article 29.	Part-ti	me Em	ployees	28
			cation of Collective Agreement	
			of Work	
	29.03		and Schedules:	
		1.	Shift Operation	
		2.	Scheduling of Shifts	
		3.	Additional Shifts	
		4.	Temporary Shift Pattern Reassignment	
		5.	Schedule Posting and Schedule Changes	
		6.	Shift Exchanges	
		7.	Reporting Pay	
		8.	Copies of Schedule	
			me	
			d Holidays	
	29.06		d Vacation	
		1.	Definition	
		2.	Vacation Entitlement	
		3.	Time of Vacation	
		4.	Vacation Pay	
	29.07	Sick I	eave	
		1.	Application of Sick h a v e	
		2.	Accrual of Sick Leave Credits	
		3.	Illness Reponing	
		4.	Sick Leave Pay Entitlement	36
		5.	Medical And Dental Appointments	
		6.	Request For Sick Credit Balance	37
		7.	Status When Sick Bank Exhausted	37

Page

A	rticle 30.	Temporary Employees
A	rticle 31.	Casual Employees
		31.01 Application of Collective Agreement
		31.02 Orientation
		31.03 Hours of Work
		31.04 Overtime
		31.05 Salaries
		31.06 Shift Differential and Weekend Premium
		31.07 Transportation
		31.08 Named Holidays
		31.09 Annual Vacation
		31.10 Dues Deductions
		31.11 Grievance Procedure
		31.12 Appointments and Transfers
		31.13 Performance Appraisal
		31.14 Discipline and Dismissal
		31.15 Shift and Schedules
	tricle 32	Layoff and Recall
	uncie 52	32.01 Joint Discussions
		32.02 Layoff Process
		32.03 Notice
		32.04 Recall
		32.05 Opportunities for Casual Work
		32.06 Benefit Coverage During Layoff
		32.07 Application of Collective Agreement
		32.08 Rights on Layoff
٨	rticle 33.	Discipline and Dismissal
	Article 33.	Resignation and Termination
л	Allicie 34.	34.01 Employee Notice of Termination
		34.02 Vacation Pay Upon Termination
	rticle 35.	Health and Safety
	vrticle 35.	Grievance Procedure
А	Anicie 50.	36,0] Grievance Definitions
		36.02 Authorized Representatives
		36.03 Time Limits
		36.04 Default
		36.05 Mandatory Conditions
		36.06 Steps in the Grievance Procedure
		36.06 Steps in the Grievance Procedure
	Article 37.	Arbitration
	rticle 38.	Copies of Conective Agreement
S	alaries Scheo	lule

Letters of Understa	nding	
1.	Optional Schedules	
2.	Booking Additional Shifts and Hours for Part-time	
	Auxiliary Nursing Staff	53
3.	Committeeto Discuss Issues of Concern	
4.	Hors of Work and Layoffs	57
5.	Administration of Educational Allowance.	
	For Nursing Attendants	
6.	Salary Appendix	60

~

COLLECTIVE AGREEMENT effective the _____ day of _____

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OF ST. ALBERT (hereinafter referred to as the "Employer")

AND

THE UNITED STEEL WORKERS OF AMERICA LOCAL 5885

(hereinafter referred to as the "Union")

PREAMBLE

WHEREAS the parties acknowledge that the primary purpose of the Employer and the Employees is **to** provide quality resident **care** and believe that this purpose can be achieved most readily if harmonious relationships exist between the Employer and the Employees;

AND WHEREAS the parties are desirous of concluding a Collective Agreement for the purpose of establishing rates of pay and other terms and conditions of employment for nursing attendants and licensed practical nurses;

NOW THEREFORE THIS COLLECTIVE AGREEMENT WITNESSES:

ARTICLE 1: TERM OF THE COLLECTIVE AGREEMENT

- 1.01 Except where otherwise stated in this Collective Agreement unless altered by mutual agreement in iti: 3 of both parties r this Ag it st be in f d effect from and after the date upon which the United Steelworkers of ri il 5885 and the Youville F it out of ratification by their is of the terms of the t, i to and including March 31, 2003 and from yx to y thereafter, unless in writing, is given by either party to the other party not less it) calendar months nor more than four (4) calendar months prior to the if its desire to amend this Agreement.
- I.02 t notice is served by either party to commence collective bargaining, this Agreement shall continue in full for at and effect until a new Collective Agreement has been executed.

ARTICLE 2:

- 2.01 "Code" means The Labour Relations d as send from time to time.
- 2.02 "Arbitration" I take the meaning from the section of the Code dealing rith the resolution of a difference.

A.D., 2000

- 2.03 "Union" means United Steelworkers of America, Local 5885. In the event of a change of name of the aforementioned Union, the subsequent name shall be recognized.
- 2.04 "Basic Rate of Pay" shall mean the step in the salary schedule applicable to the Employee, as specified in this Collective Agreement, exclusive of all premium payments or other allowances.
- 2.05 "Employee"shall mean a person covered by this Collective Agreement and employed by the Employer. At the time of hire each Employee shall be assigned by the Employer to one of the following categories: Full-time, Part-time, Casual or Temporary, and such assignment shall not be altered except in accordance with the provisions of this Collective Agreement.
 - (a) "Regular Employee" is one who works on a full-time or part-time basis on regularly scheduled shifts of a continuing nature:
 - "Full-timeEmployes" is one who is regularly scheduled to work the full specified hours in the "Hours of Work" Article of this Collective Agreement.
 - (ii) "Part-timeEmployee" is one who is regularly scheduled for less than the normal full number of hours specified in the "Hours of Work" Article of this Collective Agreement.
 - (b) "CasualEmployee" is:
 - (i) one who maybe regularly scheduled for a period of three (3)months or less for a specificjob; or
 - (ii) one who relieves for absences in a full-time or part-time position, the duration of which is three (3) months or less; or
 - (iii) one who works on a call-in basis and is not regularly scheduled.
 - (c) "TemporaryEmployee" is one who is hired on a temporary basis for a fulltime or part-time position:
 - (i) for a specific job **at more** than three (3) months but less than **six** (6) months; or
 - (ii) to replace a Full-time of Part-time Employee who is on approved leave of absence for a period in excess of three (3) months; or
 - (iii) to replace a Full-time br Part-time Employee who is on leave due to illness or injury where the Employee has indicated that the duration of such leave will be in excess of three (3) months.

- **2.06** "Employer" shall mean and include such officers as may from time to time be appointed or designated to carry out administrativeduties in respect of the operation and management of the Institution.
- 2.07 "Feminine Gender" shall mean and include the masculine and similarly, the singular shall include the plural and vice-versa, as applicable.
- 2.08 "Facility" means the health facility named as the "Employer" in this Collective Agreement.
- 2.09 "Pyramiding" means the payment of two (2) or more premiums under different provisions of the Collective Agreement for the same hours worked.
- 2.10 "Shift" shall mean a daily tour of duty excluding overtime hours.
- 2.11 "ShiftCycle" means the **period** of time when the shift schedulerepeals itself. In those instances where the scheduledoes not repeat itself, the term "ShiftCycle" shall be understood to mean a period of time not exceeding six (6) weeks.
- 2.12 For the purpose of applying the terms of this Collective Agreement, time worked shall be deemed to have been worked on the day on which the majority of hours of the shift falls.

ARTICLE 3: RECOGNITION

- 3.01 The Employer recognizes the Union as the sole bargaining agent for the Employees covered by the Collective Agreement as described in the Certificate of the Labour Relations Board and amendment? thereto.
- 3.02 No Employee shall be required to make any written or verbal agreement which is in conflict with the terms of this Collective Agreement unless the Union agrees otherwise.

ARTICLE 4: UNION MEMBERSHIP AND DUES DEDUCTION

- **4.01** Employees shall be permitted **to** wear a pin representative **of** their Union during all hours of employment.
- 4,02 Membership in the Union is voluntary.
- 4.03 An amount equal to the membership dues shall be deducted at the rate prescribed by the Union from the Employee's Basic Rate of Pay as a condition of employment.
- 4.04 The deductions shall be made by the Employer and shall be submitted to the Union not later than the 15th day of the month following and shall be accompanied by a list of **names** of those Employees from whom the deductions were made showing the amount deducted from each Employee.

- 4 -
- 4.05 The deductions above may be toke and submitted more frequently than once per month and pro-rated to the monthly dues level.
- 4.06 The dues structure of the Union shall be on a percentage basis and the Union shall give not less than thirty (30) days notice of any change in the rate at which dues are to be deducted. Any change in thear ount of deductions shall be implemented by the Employer at the next possible pay period following the expiry of the notice period.
- 4.07 The Employer shall indicate the dues deducted and enter the amount on the T-4 slips supplied to the Employee.

ARTICLE 5: HUMANITY FUND

5.01 Unless otherwise provided in Article 5.02, the Employer shall deduct from each Employee within the scope of the bargaining unit, one cent (\$.01) per hour for the purpose of International Aid and Development. A chequefor such amount shall be forwarded to the "HumanityFund" at the United Steelworkers of America National Office, accompanied by a list indicating the amount, total and names of Employees from whom deductions have been made. A copy of the list shall be provided to the Union Chairperson.

5.02 An Employee may have such deductions from their Basic Rate of Pay discontinued, by notifying the Employer and the Union of such desire in writing.

5.03 The provisions of this article shall become in force and effect as \$000 after date of ratification that is reasonably possible, and administered in a fashion consistent with the payroll system in force and effect.

ARTICLE 6: BULLETIN BOARDSPACE

6.01 The Employer shall provide a bulletin board for the exclusive use of the Union to be placed in a reasonably accessible location upon which designated space shall be provided where the Union may be permitted to post notices of meetings and other such notices which may be of interest to Employees. It is not the intention of the Union to post anything objectionable to the Employer.

ARTICLE 7: MANAGEMENTRIGHTS

- 7.01 The Union acknowledges that it shall be the exclusive right of the Employer to operate and manage the business of the Institution in all respects except as otherwise provided for in this Collective Agreement.
- 7.02 Without limiting the generality of the foregoing, the Union acknowledges that it shall be the exclusive right of the Employ of to operate and manage its business, including the right to:
 - maintain order, discipline, efficiency and to make, alter and enforce, from time to time, rules and regulations to be observed by an Employee, which are not in conflict with any provision of this Collective Agreement;

- (ii) direct the workingforce and to create new classifications and work units and to determine the number of Employees, if any, needed from time to time in any work unit or classification and to determine whether or not a position, work unit, or classification will be continued or declared redundant;
- (jii) hire, promote, transfer, lay-off and recall Employees;
- (iv) demote, discipline, suspend or discharge for just cause.

ARTICLE 8: NO DISCRIMINATION

8.01 There shall be no discrimination practised in respect of any Employee by either party as prohibited by the provisions of the Individual's Rights Protection Act, nor by reason of her membership arnon-membership or activity in the Union nor in respect of an Employee's or Employer's exercising any right conferred under this Collective Agreementor any law of Canada or Alberta. An Employee may only grieve on the basis of this clause if they have not availed themselves of the procedures available under the Individual's Rights Protection Act. Should an Employee et to pursue the remedy under that Act, such grievance shall be deemed to be abandoned.

ARTICLE 9: PROBATIONARY PERIOD

- 9.01 (a) A newly hired Regular or Temporary Employee shall serve a probationary period of five hundred and three point seventy-five (503.75) hours worked.
 - (b) The probationary period may be extended for a maximum period of two hundred and fifty-six (256) hours worked, subject to mutual agreement by the Employer, Union and Employee.
 - (c) Ha new Regular or Temporary Employee is unsuitable in the opinion of the Employer, such Employee may be terminated at any time during the probationary period without:
 - (i) notice; or
 - pay (except as may be required by the provisions of the Alberta Employment Standards Code). and
 - (iii) shall not have recourse to be grievance procedure with respect to such termination.
- 9.02 The Employer shall provide a paid orientation period for all new Employees.
- 9.03 The Employer shall provide a performance appraisal, in writing, of each probationary Employee at least once during **her** probationary period.
- 9.04 A representative of the Union shall be notified of Employee orientations and have the right to make a presentation of up to thirty (30) minutes at the orientation of new Employees. Attendance at the presentation shall not be compulsory.

ARTICLE 10: SENIORITY

10.01	(a)	An Employee's "Seniority Date" shall be the date on which a Regular or
	()	Temporary Employee last commenced employment with the Employer in the
		bargaining unit, which is uninterrupted by any occurrence outlined in Article
		10.03 below, Therefore, in accordance with the provisions of Article 31.01,
		a Casual Employee shall not be entitled to seniority, however, the seniority
		of a Regular or Temporary Employee will include prior uninterrupted service
		as a Casual Employee.

- (b) Seniorityshall not apply during the probationary period, however, once the probationary period has been completed seniority shall be credited from the date established pursuant to Article 10.01(a).
- 10.02 Seniority shall have application to:
 - (a) preference of vacation time in accordance with Article 21;
 - (b) transfers and in filling vacancies within the bargainingunit in accordance with the provisions specified in Article 11;
 - layoffs, displacement of less senior Employees, and recalls, in accordance with the provisions specified in Article 32;
 - (d) temporary shift pattern re-assignment in accordance with Article 14.03.
- **10.03** Seniorityshall **be** considered brokent and all rights forfeited, and there shall be **no** obligation to rehire:
 - (a) when the employment relationship is terminated by either the Employer or the Employee:
 - upon the expiry of twenty-four (24) months following lay-off during which time the Employee has not been recalled to work;
 - (c) if an Employee does not return to work upon recall. as provided in Article 32.04;
 - (d) if an Employee does not return from a leave of absence as scheduled except for reasons acceptable to the Employer.
- 10.04 Within three (3) months of the signing date of this Collective Agreement, and every six (6) months thereafter, the Employer shall provide to the chairperson of the Union, a seniority list containing the name and seniority date of each Regular and Temporary Employee in chronological order. The current seniority list shall also be posted on the bulletin board. The Union shall have one (1) month in which to take issue with any changed seniority dates, or the seniority date for any Employee added to the seniority list, otherwise the date for tach Employee identified on the seniority list shall stand.

!

10.05

Should a difference arise regarding an Employee's seniority. the Employer will provide the Employee with the informationnecessary to establish accurate seniority.

-7-

- 11.01 The Employer shall post, within the Facility, notices of vacancies for full-time, (a) pan-time and temporary positions covered by this Collective Agreement not less than eight (8) calendar days in advance of making an appointment. The posting shall indicate the qualifications of the position and for (b) informational purposes only, the number of hours per shift and the number of shiftsper cycle. Copies of such notices of vacancies shall be forwarded to the Union (¢) Chairperson. Applications for vacancies or transfer. shall be made in writing to such officer of the 11.02 Facility as the Employer may designate. When circumstances require that the Employer fill a vacancy before the completion 11.03 of the posting and selection process, any appointment shall be made on a temporary or casual basis only. The Employer will continue to fill the vacancy through the Parttime Availability Roster. or the casual list. When making transfers and filling vacant positions covered by this Collective 11.04 Agreement the Employer shall apply a consistent selection process. The determining factors shall be the most requisite jobrelated skills, training, knowledge, experience and other qualities, and characteristics which bear a reasonable relationship to the responsibilities and requirements of the position, and where these factors are relatively equal, seniority shall be the deciding factor. 11.05 The name of the successful candidate for transfer shall be posted for seven (7) (a) calendar days within the Facility. All other applicants for the vacancy shall be informed in writing of the successful applicant within five (5) calendar days of the appointment. If an unsuccessful $\operatorname{Employee}$ believes that she should have been selected to (b) a position pursuant to the provisions of this Article, such Employee is encouraged to meet and discuss the reasons for such decisions with her immediate supervisor. The Employee should be accompanied by a shop steward unless the Employeechooses otherwise. 11.06 A vacancy resulting from either: (a)
 - (i) the creation of a specificjob of more than three (3) months but less than six (6) months; or
 - (ii) a leave of absence granted for a period known to be longer than three (3) months, shall be posted pursuant to Article **11.01**.

- (b) (i) Where a vacancy for a temporary position has been tilled by the appointment of a Regular Full-time or Pad-time Employee, and where, at the completion *a* the expected term of the temporary position, the Employer decides that the Employee is no longer required in that position, she shall be reinstated or placed in accordance with Article 11.06. A Regular Employee achieving a temporary position shall maintain her status as a Regular Employee.
 - (ii) Where a vacancy for a temporary position has been filled by the appointment of a Casual Employee, the Casual Employee shall change her status to Temporary Employee and, where, at the completion of the temporary position, she shall be reinstated to casual status.
 - (iii) At the termination of a temporary position, the Employer shall endeavour to reinstate a Regular Employee in her former position, and if such reinstatement is not possible, place the Employee in another suitable regular position within the bargaining unit, and at a rate of pay equivalent to her former position. The reinstatement or placement of an Employee in accordance with this provision shall not be construed as a violation of the job posting or scheduling provisions.
- (c) **Daring** the term of a temporary position, the incumbent Employee shall be eligible to apply on postings in accordance with the following:
 - such Employee shall be eligible to apply on postings of vacancies pursuant to Article 11.01 (a).
 - (ii) such Employee shall not be eligible to apply on postings of vacancies pursuant to Article 11.06 (a), unless the position posted commences after the expiry of the term for which she was hired.

ARTICLE 12: ANNUAL PERFORMANCE APPRAISALS

- 12.01 The parties recognize the desirability of an or.going appraisal system designed to provide effective communications between the Employer and the Employee regarding an Employee's performance.
- 12.02 (a) A written summary of annual performance of Employees shall be completed approximately once every twelve (12) months.

- (b) Meetings for the purpose of the performance appraisal interview shall be scheduled by the Employer in consultation with the Employee. At the interview the Employee shall be given a copy of her performance appraisal. The Employee shall sign her performance appraisal for the sole purpose of indicating that she is a ware of the contents of the performance appraisal and shall have the right to respond, in writing. within seven (7) days of the interview and that her reply shall be attached to her evaluation and placed in her personnel file.
- (c) An Employee attendingher performance appraisal interview shall not suffer any loss of regular pay.
- 12.03 (a) By appointmentmade at least one (I) working day in advance an Employee may view her personnel file once each year or when the Employee has filed a grievance. An Employee may be accompanied by a Union representative when viewing her personnel file.
 - (b) An Employee may request a copy of the contents of her personnel file once in a calendar year or when the Employee has filed a grievance, provided that she first pays to the Facility a fee to cover the cost of copying.
- 12.04 An Employee's performance appraisal shall not be released by the Employer to any person except to a Board of Arbitration, or as required by law. without the written consent of the Employee.

ARTICLE 13: HOURS OF WORK

- 13.01 (a) This Article shall not be construed as a guarantee of pay or of hours of work.
 - (b) Regular hours of work for Full-time Employees, exclusive of meal periods shall be:
 - (i) seven and three quarter (7 3/4) consecutive hours per day;
 - thirty eight and threequarter (383/4) hours per week averaged over one complete cycle of the shift schedule.
- 13.02 Regular hours of work shall be deemed to:
 - (a) include, as scheduled by the Employer, two (2) rest periods of fifteen (15) minutes during each full working shift of seven and threequarter (7 3/4) hours: or
 - (b) one rest period of thirty (30) minutes during each full working shift of seven and threequarter (7 3/4) hours; if this is more compatible with scheduling of work assignments, the alternate to be applied shall be at the discretion of the Employer; or

- (c) include,as scheduled by the Employer, one (1) paid rest period of fifteen (IS) minutes during each half shift of not less than four (4) hours; and
- (d) exclude a meal period of thirty (30) minutes to be scheduled by the Employer during each working day on which the Employee works in excess of four (4) hours;
- (e) if an Employee is unable to take her meal period or rest period or is recalled to duty during her meal period or rest period she shall be given a full meal period or rest period later in her shift, or, where that is not possible, be paid for a meal period or rest period as follows:
 - for the rest period, one and one-half times (11/2X) her Basic Rate of Pay rather than at straight time; or
 - (ii) for a meal period, at one and one-half times (1 1/2X) her Basic Rate of Pay.
- 13.03 On the date fixed by proclamation, in accordance with the Daylight Savings Time Act, of conversion to Mountain Standard Time, regular hours of work shall be extended to include the resultant additional hour with additional payment due therefore at the overtime rate. On the date fixed by said Act for the resumption of Daylight Savings Time, the resultant reduction of one hour in the shift involved shall be effected with the appropriate/eduction in regular earnings.

ARTICLE 14: SHIFTSAND SCHEDULES

14.01 Shift Operation

- (a) It is recognized that the Facility operates twenty-four (24) hours per day, seven (7) days per week and three hundred and sixty-five (365)days per year. It follows, therefore, that staffing will be required on all three (3) shifts (days, evenings and nights). To ensure an appropriate efficient staffing pattern as determined by the Employerit may be required that Employees work on any or all of the three (3) shifts.
- (b) The first shift of the working day shall be the one wherein the majority of hours worked falls between twenty-four hundred (2400) and zero eight hundred (0800) hours.

14.02 Scheduling of Shifts

(a) Notwithstanding the provisions of Article 14.01(a), it is recognized and understood that the preferred shift patters, framework is permanent day, permanent evening and permanent night shifts. The Employer shall endeavour to generally maintain this pattern of shift scheduling unless it is not viable to do so.

- (b) The provisions of Article 14.02(a) shall not be deemed to prevent the Employer from accommodating Employee requests for other shift patterns when the existence of a vacancy permits the Employer to do so, subject to the provisions of Article 11, or when other affected Employees also quest to do so.
- (c) Except in cases of emergency or by mutual agreement between a Regular Employee and the Employer, shift schedules shall:
 - provide for at least fifteen and one-half (15 1/2) hours off-duty between shifts;
 - (ii) provide for at least two consecutivedays of rest except that, twice in a two (2) week cycle, there may be a singleday of rest which may be followed by not more than six (6) consecutive working days:
 - provide for days of rest on at least two (2) weekends in a five (5) week
 period. "Weekend"shall mean a Saturday and the followingSunday, assuring a minimum of fifty-six (56) hours off duty;
 - (iv) not provide for more than six (6) consecutive scheduled days of work;
 - (v) not provide for different shifts between scheduled days off.

If an Employee is required by the Employer to change shifts without receiving fifteen and one-half $(15 \ 1/2)$ hours off duty, she shall be entitled to premium pay at time and one-half $(1 \ 1/2X)$ her Basic Rate of Pay for that shift. This Section does not apply to cases where Article 14.04(c) has been applied in altering a shift schedule.

- (d) Optional scheduling provisions may be mutually agreed in writing between the Employer and the Union.
- 14.03 Temporary ShiftPattern Reassignment
 - (a) The Employer may temporarily change an Employee's shift pattern on giving at least fourteen (14) calendar days notice, in the circumstances outlined below:
 - (j) to ensure that Employees have access to inservice programs; or
 - (ii) in the event of absences for any reason, where anyone such absence exceeds fourteen (14) calendar days; or
 - (iii) to develop or maintain proficiency or knowledge as may be required; or
 - (iv) where it is considered necessary to place an Employee on the day shift to monitor or supervise performance and/or behaviour; or

- 12 -

 (v) where it is considered desirable to change an Employees shift due to the health of the Employee.

In the event of an unforeseen emergency, however, the fourteen (14) days notice shall not apply.

- (b) It is recognized that the Employer may require another Employee in the same classification to change or exchange shifts fora period not to exceed fourteen (14) calendar days due to a Full-time Employee having a temporary shift change pursuant to Article 14.03(a). Unless otherwise provided in Article 14.03(c), in such circumstances the affected Employee(s) will be selected based upon the following criteria:
 - (i) unless another Full-time Employee volunteers to make the exchange, the least seniorFull-time Employee on the applicable shift, shall be required to make the change, and
 - (ii) where no other Full-timeEmployeeexists on the applicableshifts, the Employer shall select the Part-time Employees to be affected by attempting to minimize the amount of operational disruption and to least negatively affect more seniorPart-timeEmployees. and
 - (iii) unless otherwise agreed between the Employer and the Employee, following the completion of the reassignment period or following fourteen (14) calendar days, whichever is the lesser, the Employee(s) shall be returned to their shift(s).
 - (iv) should the need as outlined in Article 14.01(a) continue to exist beyond fourteen (14) calendar days, the process outlined in (i) at:d (ii) above may be repeated, and repeated until the need has ceased to exist.
- (c) In the event that an Employee has had her shift changed pursuant to Article 14.03(b) above within the preceding six (6) months, such Employee shall he the last Employee subject to a shift change pursuant to Article 14.03(b), unless the Employee requests otherwise.
- (d) A Full-time Employee shall not have her full-time status or hours changed solely due to **the** operation of this provision.

14.04

(a) Shift schedules shall be posted twelve (12) weeks in advancesubject lo such changes as arise from application of Articles 14.02(b) or 14.03.

Schedule Posting and Schedule Changes

(b) If the Employer changes an Employee's scheduled days off, the Employee shall be paid at the rate of one and one-half times (1 1/2X) her Basic Rate of Pay for all hours worked on what would otherwise have been her off-duty days, unless fourteen (14) days notice of such change has been given. (c) If the Employer changes an Employee's scheduled shift. but not her scheduled days off. she shall be paid at one and one-half times (1 1/2X) her Basic Rate of Pay for all hours worked during the first shift of the changed schedule, unless fourteen (14) days notice of such change has been given.

14.05 Shift Exchanges

- (a) Employees may exchange shifts among themselves, provided that:
 - the exchange is agreed to, in writing, between the affected Employees; and
 - (ji) prior approval of such exchange has been given by the Employees' immediatesupervisor(s);
 - (iii) where such a request is made in writing, the Employer's reply shall also be in writing.
- (b) Such exchange shall not be deemed a violation of the scheduling provisions of this Article.

14.06 Reporting Pay

In the event that an Employeereports for work as scheduled and is requested by the Employer to return home and then return to work for a later shift, the Employee shall becompensated for the inconvenience by **a** payment equivalent to three (3) hours pay at the Employee's Basic Rate of Pay.

14.07 A copy of all Employe shift schedules shall be available to the Union Chairperson,

ARTICLE 15: OVERTIME

- 15.01 Overtime is all hours worked by an Employee, authorized by the Employer, in excess of seven and three quarter (7 3/4) hours per day, or on scheduleddays of rest.
- (a) The overtime rate of one and one-half (1 1/2X) times the Basic Rate of Pay shall be paid for work performed
 - During the first four (4) hours worked in excess of seven and threequarter (7 3/4) hours in a day.
 - (ii) On scheduleddays of rest up to seven and three-quarter (7 314)hours.
 - (b) The overtime rate of two times (2X) the Basic Rate of Pay shall be paid for work performed
 - After the first four (4) hours worked in excess of seven and threequarters (7 314) hours in a day.

- 14 -

 (ii) In excess of seven and three-quarter (7 3/4) hours worked on a scheduled day of rest or a named holiday.

ARTICLE 16: SALARIES

```
16.01 Application of Salaries Schedule
```

- (a) The basic rates of pay as set out in the Salaries Scheduleshall be applicable to all Employees covered **under** this Collective Agreement.
- (b) Unless otherwise changed by the operation of this Collective Agreement:
 - Full-time Employees shall be entitled to advance to the next higher step in the salary schedule on the successful completion of one (1) year of continuous full-time employment.
 - (ii) A Part-time Employee shall be entitled to advance to the next higher step in the salary schedule upon completion of two thousand twentytwo point seven five (2022.75) hours worked. A Part-time Employee shall be entitled to request and receive a summary of hours accrued towards her next increment once each fiscal year.
 - (iii) A Regular Part-time Employee, or Temporary Employee, or a Casual Employee who has had a change in status to a Regular Full-time Employee shall have ber anniversary date established for the purposes of (a) above based upon hours worked at the Basic Rate of Pay with the Employer at the step in the salary schedule such Employee was entitled to receive immediately prior to her change in status.

16.02 Change in Job Content

In the event that the Employer significantly changes the primary functions of a Nursing Attendant, and the Union believes that a new classification within the scope of the bargaining unit has been created, the following shall occur:

- (a) Within thirty (30) days of the date of the change, the Union shall notify the Employer. in writing, of its belief that a new classification within the bargaining unit has been created.
- (b) The Employer and Union shall endeavour to resolve the matter. If the Employer and Union cannot, ash agreement, the Union shall have thirty (30) calerdar days from the dat the Union provided notice to the Employer pursuant to Article 16.02(a), advance the question of whether a significant change in primary functions has occurred and a new classification in the bargaining unit has been chief or will refer the matter of Basic Rate of Pay to the parties pursuant to A (cle 16.03(b).

- (c) If the parties agree that a new classification has been created, the provisions of Article 16.03 shall apply.
- (d) The time limits may be extended, by mutual agreement in writing.

16.03 Creation of a New Classification

- (a) Should the Employer find it necessary to create a new classification during the term of this CollectiveAgreement, the new classification will be included within the scope of this Collective Agreement provided that:
 - (i) the parties to this Collective Agreement mutually agree that the classification is within the scope of this Collective Agreement. or failing that;
 - The Labour Relations **Board** rules that the new classification is within the scope of this Collective Agreement.
- (b) If a new classification is created under Article 16.03(a) above, for which there is no pay scale in this Collective Agreement, the Employer may establish an interim pay rate and agrees to negotiate an appropriate pay rate with the Union. Failing agreement, the parties will submit the question directly to Arbitration for settlement. The resulting pay rate shall be implemented retroactively to the date the new classification was established.

16.04 Attendance at Meetings and Special Events

Regular Employees required by the Employer to attend in-service education, staff meetings, disaster plan exercise; and committee meetings shall be recognized **as** being on duty under the terms of this Collective Agreement and an Employee attending such shall be paid at the applicable rate of pay.

ARTICLE 17: NO PYRAMIDING

17.01 There shall be no pyramiding of benefits, premiums, or other allowances, except as otherwise provided in this Collective Agreement.

ARTICLE 18: SHIFT DIFFERENTIAL AND WEEKEND PREMIUM

18.01 A shift differential of seventy-five cents (\$0.75) per hour shall be paid to an Employee working on a shift where the majority of hours of such shift fall within the period of fifteen hundred (1500) hours and zero seven hundred (0700) hours. Shift differential payment shall not be considered as part of the Employee's Basic Rate of Pay.

18.02 A weekend premium of twenty-five (\$0.25) cents per hour shall be paid in addition to shift differential, if applicable, to an Employee working a shift wherein the majority of hours of such shift fall during a sixty-four (64) hour period commencing at fifteen hundred (1500) hours on a Friday or the start of the evening shift if later than fifteen hundred (1500) hours. Such premium pay shall not be considered as part of the Employee's Basic Rate of Pay.

ARTICLE 19: TRANSPORTATION

- 19.01 RegularEmployeeswho normally travel from the Facility to their place of residence by means of public transportationfollowing the completion of their shift, but who are prevented from doing so by being required to remain on duty longer than their regular shift and past the time when normal public transportation is available, shall be reimbursed for the cost of reasonable, necessary and substantiated transportation expense from the Facility to their place of residence.
- 19.02 An Employee who is required to accompany a resident to another facility for medical reasons, shall be reimbursed for the **cost** of reasonable, necessary and substantiated transportation expenses.

ARTICLE 20: NAMED HOLIDAYS

20.01 Regular Full-time Employees shall be entitled to a day off with pay on or for the following named holidays:

New Year's Day Family Day Good Friday Victoria Day Canada Day August Civic Day Labour Day Thanksgiving Day Remembrance Day Christmas Day Boxing Day

and all general holidays proclaimed to be a statutory holiday by any of the following:

- the Municipality in which the Facility is located, but only where such civic holiday is in lieu of the August Civic Holiday; or
- (b) the Province of Alberta; or
- (c) the Government of Canada.

20.02

- To qualify for a named holiday with pay the Employee must:
 - (a) work her scheduled shift immediately prior to and immediately following the holiday, except where the Employee is absent due to illness or other reasons acceptable to the Employer; and
 - (b) work on the holiday when scheduled or required to do so.

17 -

- (c) Notwithstanding the foregoing, while:
 - (i) **on** layoff; or
 - (ii) in receipt of compensation from the Workers' Compensation Board; or
 - (iii) on unpaid absenceduring which she is in receipt of weekly indemnity as provided for by the Shoht Term Disability Income Insurance Plan or the Long Term Disability Income Insurance Plan; or
 - (iv) on other leaves of absence for any reason in excess of thirty (30) calendar days.

An Employee shall not be entitled to:

- (v) a day off with pay, or
- (vi) payment in lieu thereof, for the aforementioned named holidays.
- 20.03 Subject to Article 2.12 an Employee required by the Employer to work on a named holiday shall be paid for all hours worked on such named holiday at one and one-half times (1 1/2X) her Basic Rate of Pay plus!
 - (a) an alternate day off at a mutually agreed time: or
 - (b) by mutual agreement, the day added to her next vacation; or
 - (c) failing mutual agreement within thirty (30) calendardays following the named holiday of the option to be applied, the Employee shall receive pay for such day at her Basic Rate of Pay.
- 20.04 When a named holiday falls on a day that would otherwise **be** a Regular Employee's regularly scheduled day off, the Employee shall receive an alternate day off as outlined in Article 20.03 above.
- 20.05 When a named holiday falls within an Employee's vacation, the Employee will be entitled to an additional day in lieu thereof. The additional day shall be added to the Employee's vacation period. and the Employee shall be paid at the Basic Rate of Pay.
- 20.06 Upon receiving a request from an Employee, where possible, the Employee shall be given either Christmas Eve and Christmas Day or New Year's Eve and New Year's Day off.

ARTICLE 21: ANNUAL VACATION

21.01 Definition:

For the purpose of this Article:

- 18 -
- (a) "Vacation" means annual vacation with pay.
- (b) "Vacation Year" means the twelve month period commencing on the first day of April in each calendaryear and concluding on the last day of March of the following calendar year.
- (c) "Date of Employment" means "date of hire".

21.02 Vacation Entitlement

- (a) During each year of employment with the Employer as a Full-time Employee, a Full-time Employee shall earn entitlement to a vacation with pay to be taken in the next following vacation year. The rate at which vacation is earned shall be based upon her total length of continuous employment as outlined below:
 - during the first (1st) year of continuous employment an Employee earns a vacation of tet (10) working days;
 - (ii) during the second (2nd) to fifth (5th) years of continuous employment an Employeeearns a vacation of fifteen (15) working days;
 - during the sixth (6th) to fourteenth (14th) years of continuous employment an Employee earns a vacation of twenty (20) working days;
 - (iv) during the fifteenth (15th) to twenty-second (22nd) years of continuous employment an Employee earns a vacation of twenty-five (25) working days;
 - (v) during the twenty-third (23rd) and subsequent years of continuous employment, an Employee earns a vacation of thirty (30) working days.
- (b) A Full-timeEmployee with less than one year of continuous employmentas a Regular or TemporaryEmployee prior to the first (1st) of April in any one (1) year shall be entitled to a vacation calculated on the number of months from the date of employment, in proportion to which the number of months of the Employee's service bears to twelve (12) months.
- (c) A Full-time Employee with less than one (1) year of full-time service prior to the first of April in any one (1) year, since clanging status from a Part-time Employee, shall be entitled to any unused vacation with pay calculated pursuant to Article 29.06(4) in addition to any vacation accrued pursuant to Article 21.02(b).

21.03 Cessation of Vacation Accrual

- (a) Notwithstanding Article 21.02 above, accrual of vacation entitlement will cease during a period of Employee absence in excess of thirty (30) calendar days, for any or a combination of the following reasons:
 - (i) illness or injury, unless in receipt of sick leave with pay pursuant to Article 24;
 - (ii) layoff;
 - (iii) leaves of absence without pay, except for a leave of absence for Union business.
- (b) Vacation benefits will accrue during the remainder of the vacation year, proportionate to the period worked.

21.04 Time of Vacation

- (a) Except as provided in Article 21.04(e), vacation earned in one vacation year shall be taken during the next vacation year. Subject to the provisions of Article 21.04(b), the Employer shall endeavour to grant an Employee's vacation at a mutually agreeable time. In the event that mutual agreement cannot be reached, the Employer shall be responsible for scheduling such vacation.
- (b) The Employer shall post the vacation schedule planner by January 1st of each year. Where an Employee submits her vacation preference by March 15th of that year, the Employer shall indicate approval or disapproval of that vacation request in writing by April 30th of the same year. Where the number of Employees indicating a preference for a specific period exceeds the number of Employees as determined by the Employer that can be allocated vacation during that period, seniority shall be the deciding factor.
- (c) For vacation requests submitted after April 30th the Employer shall indicate approval or disapproval within fourteen **(14)** days.
- (d) A vacation period may be divided by mutual agreement between the Employee and the Employer.
- (e) Once a vacation period is scheduled and approved, it shall not be changed while the Employee remains in the same position and unit unless mutually agreed by the Employer and the Employee. For the purposes of this clause, a vacation period is a period of t ne away from the workplace in calendar days. This clause is intended solely to ensure that rescheduling of vacation does not cause inconvenience to (he Employer or the Employee. Therefore, this clause shall not restrict the application of any other provisions of the Collective Agreement (example: making changes to the shift schedule), nor shall it serve to increase the Employee's entitlement to vacation with pay.

(f) An Employee may be permitted to carry forward a portion of unused vacation to the next vacation year. Such requests shall be made in writing and shall be subject to the Employer's approval.

ARTICLE 22: EMPLOYEE BENEFITS

- 22.01 When the enrollment and other requirements of the insurer(s) have been met, the Employer shall take steps to contractifor and implement the following group plans:
 - (a) Alberta Blue Cross Supplementary Benefits Plan, or equivalent, eighty percent (80%)Reimbursement Drug Benefit List Benefits Plan; or equivalent;
 - (b) The Health Organizations Benefits Plan, Dental Plan or equivalent, which provides for the reimbursement of eighty (80%) of eligible Basic Services; fifty percent (50%) of all eligible Extensive Services; and fifty percent (50%) of eligible Orthodontic Services in accordance with the current Alberta Dental Association Fee Guide. A maximum annual reimbursement of fifteen hundred dollars (\$1500) per insured person per benefit year shall apply to Extensive Services. Orthodontic Services shall be subject to a lifetime maximum reimbursement of fifteen hundred dollars (\$1500) per insured person.
 - (c) Alberta Health Care Insurance Plan:
 - (d) Group Life Insurance, including basic Accidental Death and Dismemberment;
 - (e) At the Employer's option, # "EI SUB Plan" to supplement an eligible Employee's Unemployment Hsurance to meet the Employer's obligation to provide benefit payments during the valid health-Elated period for being absent from work due to pregnancy for which she has provided satisfactory medical substantiation.
 - (f) Long Term Disability (income replacement during a qualifying disability equal to sixty-six and two-thirds percent (662/3%) of basic monthly earnings at the Basic Rate of Pay to the established maximum following a one hundred and twenty (120) working day elipination period).
- 22.02 Where the benefits specified above are provided through insurance obtained by the Employer, the administration of such plans shall be subject to and governed by the terms and conditions of the policies or contracts entered into with the underwriters of the plans.
- 22.03 The premium costs of the above plans listed in Article 22.01 shall be shared seventy percent (70%) by the Employer and thirty percent (30%) by the Employee.
- 22.04 The Employer shall distribute **brochures** and other relevant information concerning the above plans to Employees at the time of hire and when changes to the plans occur.

- 22.05 The Employer shall provide one copy of each of the plans to the Provincial Office of the Union.
- 22.06 Such coverage shall be provided to Regular and Temporary Employees except for:
 - a Regular Part-time or Temporary Employee whose hours of work are less than fifteen (15) hours per week averaged over one complete cycle of the shift schedule; and
 - (b) a Temporary Employeewho is hired for a position of less than six (6)months. who is eligible to participate only in Article 22.01(a) and (b) above.

ARTICLE23: PENSION PLAN

- 23.01 The Employer shall continue the Pension Plan in existence as at the date of ratification to provide benefits for participating Employees, provided they are scheduled to work at least fifteen (15) hours per week as averaged over one (1) complete cycle of the shift schedule, in accordance with the terms and conditions of the Plan.
- 23.02 Contributions to the Plan shall be shared equally between the Employer and eligible Employees.
- 23.03 The Employer shall distribute brochures and other relevant material outlining the Plan to eligible Employees at the time of hire and when there are changes to the Plan.

ARTICLE 24: SICK LEAVE

24.01 Application of Sick Leave

- (a) Sick leave is provided by the Employer for absences due lo illness, quarantine by a Medical Officer of Health, or because of an accident for which compensation is not payable under the Workers' Compensation Act.
- (b) The Employer recognizes that alcoholism, drug addiction and mental illness, are illnesses which can respond to therapy and treatment, and that absences from work due to such therapy shall be considered sick leave.

24,02 Accrual of Sick Leave Credits

(a) After Regular or Temporary Employees have completed five hundred and three point seven five (503.75) hours worked she shall be allowed a credit for sick leave computed from the date she last commenced employment as a Regular or Temporary Full-time Employee at the rate of one and one-half (1 112) working days for each full month of employment up to a maximum credit of one hundred and twenty (120) working days.

(b) When an Employee has accrued the maximum sick leave credits of one hundred and twenty (120) working days she shall no longer accrue sick leave credits until such time as her total accumulation is reduced below the maximum. At that time she shall recommence accumulating sick leave credits.

24.03 Cessation of Sick Leave Credit Accrual

- (a) Sick leavecredits shall **not** accrue during a period of Employee absence in excess of thirty (30) calendar days for any or a combination of the following reasons:
 - (i) illness or injury;
 - (ii) layoff;
 - (iii) leaves of absence without pay, except for a leave of absence for Union business.

24.04 Illness Reporting

Employees reporting sick shall advise the Employer **as soon** as possible and regularly thereafter **as** required by the Employer.

24.05 Sick Leave Pay Entitlement

- (a) An Employee shall not be entitled to apply sick leave credits prior to the completion of five hundred and three point seven five (503.75) hours worked since she last commenced employment as a Regular or Temporary Employee.
- (b) An Employee granted sick leth/e shall be paid for the period of such leave at her Basic Rate of Pay and t^{Re} number of days thus paid shall be deducted from her accumulated sick leave credits up to the total amount of the Employee's accumulatedcre⁴/₄ at the time the sick leave commenced.
- (c) Employees may be required to submit proof satisfactory to the Employer of any illness, non-occupational accident or quarantine. Payment of sick leave benefits shall not be effected until required substantiation has been supplied.
- (d) No sick leave shall be granted for any illness or injury which is incurred once an Employee commences her vacation; in this event, the Employee will be receiving vacation pay. Notwithstanding the foregoing, should an Employee demonstrate to the satisfaction of the Employer that she was admitted to a hospital as an "in-patient" configuous to or during the course of her vacation, she shall be considered to be on sick leave for the period of the stay in the hospital, subject to the provisions of Article 24.05(b). Vacation time not taken as a result of such stay in the hospital shall be taken at a mutually agreeable later date.

24.06 Medical and Dental Appointments

If an Employee requires **time off for** the purpose of attending a dental, physiotherapy. optical or medical appointment, provided she has been given prior authorization by the Employer, such absence shall be charged against her accumulated sick leave credits. Employees may be required to submit satisfactory proof of such appointment.

24.07 Request For Sick Credit Balance

Upon request of an Employee, but not not need a year, the Employer shall advise an Employee of her accrued sick leave credits.

24.08 Status When Sick Rank Exhausted

- (a) If an Employeehas exhausted her sick leave credits during the course of an illness, and the illness continues. she shall be deemed to be on leave of absence without pay or benefits except as provided in Article 27.01(c), for the duration of the illness or as provided below. The Employee shall keep the Employer advised as to when she may be expected back to work and shall provide the Employer with two (2) weeks written notice of her readiness to return to wark. The Employer shall then reinstate the Employee in the same position held by her immediately prior to the disability with the remaining benefits that accrued to her prior to the disability. At the expiration of twenty-four (24) months from the last day of paid sick leave, an Employee who is not capable of returning to work pursuant to this Article shall be considered to have terminated her employment relationship with the Employer, provided that such termination is not contrary to any right conferred under this agreement or any law of Canada or Alberta.
- (b) The reinstatement of an Employee in accordance with this Article shall not be construed as being a violation of the posting and/or scheduling provisions of Articles 11 and 14.

ARTICLE 25: WORKERS COMPENSATION

- 25.01 Each accident or injury incurred on duty **must** be reported to the Employer.
- 25.02 Employees shall be covered by the Workers' Compensation Act and Regulations thereunder. An Employee who is unable to work as a result of a disability incurred while on duty in the service of the Employerand who qualifies for benefits pursuant to the Workers' Compensation Act, will receive compensation benefits directly from the Workers' Compensation Board. U the Employee incurs delays in receiving payment(s) from the Workers' Compensation Board, the Employer will advance or lend money in the amount(s) due from the Workers' compensation Board provided the necessary repayment or cheque transfer formsare signed.

- 24 -
- 25.03 An Employee who has been on Workers' Compensation and who is certified by the workers' Compensation Board to be tit to return to work and who is capable of performing the dudes of her former position shall provide the Employer with two (2) weeks written notice of her readiness to return to work. The Employer shall then reinstate the Employee in the same position held by her immediately prior to the disability with benefits that accrued to her prior lothe disability.
- 25.04 At the expiration of twenty-four (24) months from the first day of absence as a result of an accident sustained while on duty in the service of the Employer within the meaning of the Workers' CompensationAct, an Employee who is not capable of returning to work shall be considered to have terminated her employment relationship with the Employer.
- 25.05 For the purpose of determiningsalary increments, a Full-time Employee who is in receipt of Workers' Compensation benefits shall be deemed to remain in the continuous service of the Employer.
- 25.06 The reinstatement of an Employee in accordance with this Article shall not be construed as being in violation of the posting and/or scheduling provisions of Articles 11 and 14.
- 25.07 Employees who are in receipt of Workers' Compensation shall not be entitled to named holidays and shall cease to accrue vacation and sick leave credits if such absence due to disability exceeds thirty (30) calendar days.

ARTICLE 26: UNIFORMS

26.01 The nature, colour and style **cf uniforms c** work attire, and the requirements in respect thereto, shall be determined by the Employer. Any monetary assistance extended by the Employer **as** at **the date** of ratification shall continue at four **(4)** cents per hour.

1

ARTICLE 27: LEAVES OF ABSENCE

27.01 General Leaves

(a) Leave of absence without pay may be granted to an Employee at the discretion of the Employer and the Employee shall not work for gain during the period of leave of absence except with the express consent of the Employer. A leave of absence without pay may be granted for education and professional or educational meetings at the discretion of the Employer.

- (b) Requests for a leave of absence, without pay or benefit of Employer contributions will, when possible, be made in writing to that Employee's immediate supervisorsix (6)weeks in advance except that in extenuating circumstances the time factor may be waived or reduced. Recognizing that the primary commitment of the Employee is to the Employer, the granting of leaves of absence is subject to the approval of the Employer. Except in exceptional circumstances the Employer will reply in writing to a request for leave of absence within fourteen (14) days of receipt of the request.
- (c) Except as provided in Article 27.01(d) where an Employee is granted a leave of absence of more than thirty (30) calendar days duration, and that Employee is covered by any or all of the plans specified in Article 22, the Employee may, subject to the Insurer's requirements, make prior arrangements for the prepayment of the full premiums for the applicable plans.
- (d) For the portion of Maternity Leave during which an Employee has a valid health-relatedreason for being absent from work and who is in receipt of sick leave or El SUB Plan Benefits, benefit premium payments shall be administered in the same fashion as an Employee absent due to illness.
 - (i) Where the Employee meets eligibility requirements for the SUB Plan and submits all of the required forms to the Employer prior to or on the payroll cut off date, they will receive SUB Plan payments on the first regular pay day following the payroll cut off date.
- (e) With the exception of a leave of absence for Union business, in the case of a leave of absence in excess of thirty (30) calendardays, Full-time Employees shall cease to accrue sick leave and earned vacation to the extent that such leave exceedsthirty (30) calendar days. A Full-time Employee's eligibility date for advancement to a higher step on the salary schedule shall also be adjusted by the same amount of time.
- (f) Where an Employee is elected or appointed to represent the Union at Conventions, Workshops, Institutions, Seminars, to attend meetings as a member of the Union's Local Executive Board or negotiations with the Employer, such leave shall not be unreasonably denied.
- (g) One Employee who is elected to a full-time position with the Union shall be granted leave of absence without pay and without Loss of seniority for a maximum period of two (2) years. Such leave of absence shall be renewable for a further term upon request. If it is permissible under the pension and group life plan and any other welfare plans, the Employee shall have the right to pay the full costs including the Facility's share, during the period of such leave of absence.
- (h) Employees shall not be entitled to named holidays with pay which may fall during the leave of absence.

(i) An Employee who has been granted a leave of absence and overstays the leave without permission of the Employer, shall automatically terminate her position except in cases of extenuating circumstances acceptable to the Employer.

27.02 Maternity Leave

- An Employee who has completed nine (9) months' continuous employment (a) shall, upon her written request, be granted maternity leave to become effective twelve (12) weeks immediately preceding the expected date of delivery, or such shorter period as maybe mutually agreed upon between the Employer and Employee, provided, however, that where in the opinion of the Employer her ability to carry out her normal work assignment becomes limited, she may be placed on maternity leave earlier. Where possible, the Employee shall advise the Employer of her intended commencement date of maternity leave fourteen (14) calendar days in advance, but in any event, shall give the Employer an estimated commencement date no later than six (6) weeks prior to the estimated date of delivery. Such leave shall not exceed nine (9) months beyond the date of delivery. Maternity Leave shall be without pay and benefits except for the portion of Maternity Leave during which the Employee has a valid health-Elated reason for being absent from work and is also in receipt of sick leave or EI SUB Plan benefits.
- (b) An Employee on such leave shall provide the Employer with fourteen (14) days notice of readiness to return to work, following which the Employer will reinstate her in the same position held by her immediately prior to taking leave, and at the same step in the pay scale, or provide her with alternate work of a comparable nature at not less than the same step in the pay scale and other benefits that accrue to her up to the date she commenced leave.
- 27.03 Adoption Leave
 - (a) An Employee who has completed nine (9) months' continuous employment with the Employer shall upon her written request. be granted leave without pay for up to nine (9) months as necessary for the purpose of adoptinga child.
 - (b) The Employee may commence adoption leave upon one day's notice provided that application for such leave is made when the adoption has been approved and the Employer is kept informed of the progress of the adoption proceedings.
 - (c) An Linployee on such leave shall provide the Employer with fourteen (14) days' notice of readiness to return to work, following which the Employer will reinstate her in the same position held by her immediately prort to taking leave, and at the same step in the pay scale, or provide her with alternate work of a comparable nature at not less than the same step in the pay scale and other benefits that accrue to her up to the date she commenced leave.

27.04 Court Appearance

An Employee required by law to appear in court as a member of a jury or as a witness, shall be paid the difference between *the* pay received from such court service and pay the Employee would have normally received if she had been working, based on her Basic Rate of Pay. The Employee will report to work during **trose**, hours that she is not required to attend court, if it is reasonable to do so.

27.05 Bereavement Leave

- (a) Bereavement leave of five (5) calendar days without loss of income shall be granted in the event of death of a member of the Employee's immediate family, i.e. children, parents, brothers, sisters, spouse (including common-law spouse), grandparents, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, sister-in-law, grandchild, fiance, or guardian. Step parent, stepchildren, stepbrother and step-sister shall be considered as members of the Employee's immediate family. Such days maybe taken only in the period which extends from the date of death to the date of interment. Bereavement leave may include normal days off and/or vacation, but no additional payment is due thereof.
- (b) The Employee shall be paid for het scheduled hours at her Basic Rate of Pay, provided that she is not otherwise entitled to payment for a named holiday,
- (c) In the event of death of a member of the Employee's immediate family, as defined in Article 27.05(a), while an Employee is on a scheduled vacation the Employee will be entitled to be eavement leave in accordance with Article 27.05(a) and the Employee, may reschedule their vacation days interrupted by the bereavement in accordance with Article 21.04.
- (d) Bereavement leave without pay as may be appropriate may be granted upon request in the event of death of other relatives or close friends.

27.06 Special Leave

if an Employee is unable to report to work as a result of illness in the immediate family requiring the Employee's personal attention, she shall inform the Employer of such with as much advancenotice as possible and at the Employee's option she shall use either a vacation day, a day earned in lieu of a named holiday, banked overtime or an unpaid leave of absence for the hours not worked. An Employee shall be entitled to take three (3) working days of special leave in a twelve (12) month period commencing April 1, 1993. The Employee may be required to submit satisfactory proof of illness. For the purpose of this article, immediate family is defined as spouse and children.

ARTICLE 28: INSERVICE EDUCATION

- 28.01 The parties to the Collective Agreement recognize the value of continuing inservice education for Employees, and that the responsibility for such continuing education lies with both the Employee and the Employee. For the purpose of this Article, the term "inservice" includes orientation, equisition and maintenance of essential skills and other programs which may be offered by the Employer.
- 28.02 The Employer reserves the right to identify specific inservice sessions as being compulsory for Employees and those required to attend shall be paid at the applicable rate of pay.

ARTICLE 29: PART TIME EMPLOYEES

29.01 Application of Collective Agreement

All provisions of this Collective Agreement shall apply to Part-time Employees, except:

Article 13 - Hours of Work Article 14 - Shifts and Schedules Article 15 - Overtime Article 20 - Named Holidays Article 21 - Annual Vacation Article 24 - Sick Leave

which are superseded by the following:

29.02 Hours of Work

- (1) (a) This provision shall not be construed as a guarantee of pay or of hours of work.
 - (b) Regular hours of work, exclusive of meal periods, shall be up to Seven and three-quarter (7 3/4) consecutive hours in any day. The ratio of work days to non-work days shall be less than 5:2 as averaged over one complete cycle of the shift schedule.
- (2) Regular hours of work shall be deemed to:
 - (a) include, as scheduled by the Employer, two (2) rest periods of fifteen
 (15) minutes each dyring each full working shift of Seven and threequarter (7 3/4) hours; or
 - (b) one (1) rest period of thirty (30) minutes during each full working shift of seven and three-quarter (7 3/4) hours; if this is more compatible with scheduling of work assignments, the alternate to be applied shall be at the discretion of the Employer; or

- (c) include, as scheduled by the Employer, one (1) rest period of fifteen (15) minutes during each half shift of not less than four (4) hours; and
- (d) exclude a meal period of thirty (30) minutes to be scheduled by the Employer during each working day on which the Employee works in excess of four (4) hours;
- (e) if an Employee is unable!to lake her meal period or rest period or is recalled to duty during her meal period or rest period she shall be given a full meal period or rest period later in her shift, or, when that is not possible, be paid for a meal period or rest period as follows:
 - (i) for the rest period, at one and one-half times (1 1/2X) her Basic Rate of Pay rather than at straight time; or
 - (ii) for a meal period, at one and one-half times (1 1/2X) her Basic Rate of Pay.
- (3) On the date fixed by proclamation, in accordance with the *Daylight Savings* Time Act, of conversion to Mountain Standard Time, regular hours of work shall be extended to include the resultant additional hour with additional payment due therefore at the overline rate. On the date fixed by said Act for the resumption of Daylight Savings Time, the resultant reduction of one hour in the shift involved shall be affected with the appropriate deduction in regular earnings.

29.03 Shifts and Schedules

(1) Shift Operation

- (a) It is recognized that the Facility operates twenty-four(24) hours per day, seven (7) days per week and three hundred and sixty-five (365) daysper year. It follows, that staffing will be required on all three (3) shifts days, evenings and nights. To ensure an appropriate efficient staffing p a 1 m as determined by the Employerit may be required that Employees work on any or all of the three (3) shifts.
- (b) The first shift of the working day shall be the one wherein the majority of hours worked falls between twenty-four hundred (2400) and zero eight hundred (0800) hours.

(2) Scheduling of Shifts

(a) Notwithstandingthe provisions of Article 29.03(1)(a), it is recognized and understood that the preferred shift pattern framework is permanent days, permanente vening and permanent night shifts. The Employer shall endeavour to generally maintain this pattern of shift scheduling unless it is not viable to do so.

- 30 -
- (b) The provision of Article 29.03(2)(a) shall not be deemed to prevent the Employer from accommodating Employee requests for other shift patterns when the existence of a vacancy permits the Employer to do so, subject to the provisions of Article 11, or when other affected Employees also request to do so.
- (c) Except in cases of emergency or by mutual agreement between a RegularEmployee and the Employer, shiftschedules shall:
 - (i) provide for at least fifteen and one-half (15 1/2) hours off duty between shifts;
 - schedule an Employee not to work on at least two (2) weekends in a five (5) week period. "Weekend" shall mean a Saturday and the following Sunday, assuring a minimum of fifty-six (56) hours off duty;
 - (iii) not provide more than six (6) consecutive scheduled days of work more that twice in a five (5) week cycle;
 - (iv) not provide for different shifts between scheduled days off.

If an Employee is **required** by the Employer to change shifts without receiving fifteen and pre-half (15 1/2) hours off duty, she shall be entitled to premium pay at one and one-half times (1 1/2X) her basic rate for that shift. This Article does **not** apply **to cases** where Article 29.03(5)(b) has been **applied** in altering**a** shift schedule.

- (d) Optional scheduling provisions may be mutually a p e d in writing between the Employer and the Union.
- (3) Additional Shifts
 - (a) A Part-time Employee may work additional shifts.
 - (b) Where a Part-time Exployee volunteers or agrees to work additional shifts. she shall be paid her basic rate for such hours, or if applicable, at the overtime rate provided in Article 29.04:
 - (i) for those **hours** worked in excess of Seven and three-quarter (7 3/4) hours in a **day**; or
 - (ii) for work performed by the Employee on days in excess of the work ratio referred to in Article 29.02(1)(b).
 - (c) Where the Employer requires a Part-time Employee to work without her having volunteered or agreed to do so, she shall be paid the overtime rate provided in Article 29.04.

- 31 -

- (d) An effort shall be made by the Employer to ensure fair and equitable distribution of extra shifts through the use of the Part-time Availability Roster.
- (4) Temporary Shift Pattern Reassignment
 - (a) The Employer may temporarilychange an Employee's shift pattern on giving at least fourteen (14) calendar days notice, in the circumstances outlined below:
 - (i) to ensure that Employeeshave access to in-service programs;
 - (ii) in the event of absences for any reason, where any one such absence exceeds fourteen (14) calendar days; or
 - (iii) to develop or maintain proficiency or knowledge as may be required; or
 - (iv) where it is considered necessary to place an Employee on the day shift to monitor or supervise performance and/or behaviour; or
 - (v) where it is considered desimble to change an Employee's shift due to the health of an Employee,

In the event of unforeseen emergency, however, the fourteen (14) days notice shall not apply.

(b) It is recognized that the Employer may require another Employee in the same classification to change or exchange shifts due to a Part-time Employee having a temporary shift change pursuant to Article

29.03(4)(a). Thless otherwise provided in Article 29.03(4)(c), in such circumstances the affected Employee(s) will be selected based upon the following criteria:

- unless another Part-time Employee volunteers to make the exchange, the least senior Part-time Employee on the applicableshift who works equivalenthours, shall be required to make the change, and
- (ii) where no Part-time Employee works equivalent hours to the Employee being reassigned, the Employer shall select the Employee(s) to be affected by attempting to minimize the amount of operational disruption and least negatively affect more senior Employees, and

- (iii) unless otherwise agreed between the Employer and the Employee, following the completion of the reassignment period or following the fourteen (14) calendar days, whichever is the lesser, the Employee(s) shall be returned to their former fift(s).
- (iv) should **the need** as outlined in Article 29.04(1)(a) continue to exist beyond fourteen (14) calendardays, the process outlined in (i) and (ii) above may be repeated, and repeated until the need has ceased to exist.
- (c) In the event that an Employee has had her shift changed pursuant to Article 29.03(4)(b) above within the preceding six (6) months, such Employe. shall be the last Employee subject to a shift change pursuant to Article. 29.03(4)(b), unless the Employee requests otherwise.
- (d) Unless otherwise agreed by the Employer and the Employee, a Parttime Employeeshall not have the number of hours she was scheduled to work prior to the change, significantly reduced solely due to the application of these provisions.
- (5) Schedule Posting and Schedule Changes
 - (a) Shift schedules shall be posted twelve (12) weeks in advance subject to such changes as arise from application of Article 29.03(2)(b) or 29.03(4).
 - (b) If the Employer changes an Employee's scheduled shift, she shall be paid at one and one half times (I 1/2X) her Basic Rate of Pay for all hours worked during the first shift of the changed schedule, unless fourteen (14) days notice of such change has been given.
- (6) Shift Exchanges
 - (a) Employees may exchange shifts among themselves, provided that:
 - (i) the exchange is agreed to, in writing, between the affected Employees, and
 - (ii) prior approval of such exchange has been given by the Employee: immediatesupervisor(s);
 - (iii) where such a request is made in writing, the Employer's reply shall also be in writing.
 - (b) Such exchange shall not be deemed a violation of the scheduling provisions of this Article.

- 33 -

(7) Reporting Pay

In the event that an Employee reports for work as scheduled and is requested by the Employer to return home and then return to work for a later shift, the Employee shall be compensated for her inconvenience by a payment equivalent to three (3)hours' pay at the Employee's Basic Rate of Pay.

(8) Copies of Schedule

A copy of all Employee shift schedules shall be available to the Union President.

29.04 Overtime

- (a) The overtime rate of one and one-half times (11/2X) or two times (2X) the Employee's Basic Rate of Pay shall be paid for work authorized by the Employer and performed by the Employee on days in excess of the work ratio referred to in Article 29.02 above; or for those hours which exceed seven and three-quarter (7 3/4) hours worked in any given day.
- (b) The overtime rate of one and one-half limes (1 1/2X) times the Basic Rote of Pay shall be paid for work performed:
 - during the first four (4) hours worked in excess of seven and threequarters (7 3/4) in a day.
 - (ii) on days in excess of the work ratio referred to in Article 29.02,
- (c) The overtime rate of two 'imes (2X) the Basic Rate of Pay shall be paid for work performed
 - after the first four (4) hours worked in excess of seven and threequarter (7 314) hours in a day.
 - (ii) in excess of seven and three-quarter (7 3/4) hours worked on a named holiday.

29.05 Named Holidays

- (a) A Part-time Employee required to work on a named holiday shall be paid at one and one-half times (1 1/2X) her Basic Rate of Pay for all hours worked on such day.
- (b) A Part-time Employeeshall be paid in addition to her Basic Rete of Pay four point two percent (4.2%) of this rate per pay period in lieu *cf* the named holidays.

- 34 -
- (c) Upon receiving a request from an Employee, where possible, the Part-time Employee shall be given either ChristmasEve and Christmas Day or New Year's Eve and New Year's Day off.
- 29.06 Annual Vacation

(1) Definition:

For the purpose of this Article;

- (a) "Vacation" means annual vacation with pay.
- (b) "Vacation Year" means the twelve month period commencing on the first day of April in each calendaryear and concluding on the last day of March of the following calendaryear.
- (c) "Date of Employment" means an Employee's date of hire.
- (2) Vacation Entitlement

Part-time Employees shall be entitled to receive time off for vacation purposes based on the number of years of continuous employment as outlined below, and shall receive vacation pay in accordance with Article 29.06(4):

- (a) during the first (1st) year of employment an Employee accumulates vacation time of fourteen (14) calendar days;
- (b) during the second (2nd) to fifth (5th) years of employment an Employee accumulates vacation time of twenty-one (21) calendar days;
- during the sixth (6th) to fourteenth (14th) years of employment an Employee accumulates vacation lime of twentyeight (28) calendar days;
- (d) during the fifteenth (15th) to twenty second (22nd) years of employment an Employee accumulates vacation time of thirty-five (35) calendar days;
- (e) during the twenty-third (23rd) and subsequent years of employment an Employee accumulates vacation time of forty-two (42) calendar days.

- (3) Time of Vacation
 - (a) Vacation earned in one (1) vacation year shall be taken during the next following vacation year. As far as possible, Regular Employees shall be granted their choice of vacation periods, however, the final allotment of vacation remains the responsibility and within the authority of the Employer.
 - (b) Vacation leave will be deemed to have commenced on the first regularly scheduled work day absent on vacation leave and continue on consecutive calendardays until return to duty. A vacation period may be divided by mutual agreement between the Employee and the Employer.
 - (c) The Employer will post the vacation schedule planner by January 1st of each year. Where an Employee submits her vacation preference by March 15th of that year, the Employer shall indicate approval or disapproval of that vacation request by April 30th of the same year.
- (4) Vacation Pay

Vacation pay is to be paid to a Part-time Employee, at least one day and not more than two weeks before the commencement of the Employee's annual vacation, and shall be in accordance with the following formula. The hours worked during the preceding vacation year and **periods** of sick leave with pay multiplied by the Basic Rate of Pay in effect on the date vacation leave commences multiplied by the applicable **rate** of:

- (a) 4% during the first (1st) employment year; or
- (b) 6% during the second (2nd) to fifth (5th) employment years; or
- (c) 8% during the sixth (6th) to fourteenth (14th) employment years; and
- (d) 10% during the fifteenth (15th) to twenty-second (22nd) employment years; and
- (e) 12% during the twenty-third (23rd) and subsequent employment years.

(example 500 hours x \$10.31 per hour x .06 = \$309.30)

29.07 Sick Leave

- (1) Application of Sick Leave
 - (a) Sick leave is provided by the Employer for absences due to illness, quarantine by a Medical Officer of Health, or because of an accident for which compensation is not payable under the Workers' Compensation Act.

- (b) The Employer recognizes that alcoholism, drug addiction and mertal illness, are, illnesses which can respond to therapy and treatment. and that absences from work due to such therapy shall be considered sick leave.
- (2) Accrual of Sick Leave Credits
 - (a) After a Regular or Temporary Employee has completed five hundred and three point seventy-five (503.75) hours worked she shall be allowed a credit for sick leave computed from the date she commenced employment as a Regular or Temporary Part-time Employee at the rate of one and one-half (1 1/2) days per month, pro rated on the basis of hours worked by the Part-time Employee, in relation to the regularly scheduled hours for a Full-time Employee, up to a maximum of one-hundred twenty (120) working days.
 - (b) When a Part-time Employee has accrued the maximum sick leave credits of one hundred and twenty (120) working days she shall no longer accrue sick leave credits until such time as her total accumulation is reduced below the maximum. At that time she shall recommence accumulating sick leave.
- (3) Illness Reporting

Pan-time Employees reporting sick shall advise the Employer as soon as possible and regularly thereafter as required by the Employer.

- (4) Sick Leave Pay Entitlement
 - (a) An Employee shall not be entitled to apply sick leave credits prior to the completion of five hundred and three point seven five (503.75) hours worked.
 - (b) A Part-time Employee granted sick leave shall be paid for the regularly scheduled shifts absent due to such leave at her Basic Rate of Pay and the number of days thus paid shall be deducted from her accumulated sick leave credits up to the total amount of the Employee's accumulated credits at the time the sick leave commenced.
 - (c) Part-time Employees may be required to submit proof satisfactory to the Employer of any illness, non-occupational accident or quarantine. Payment of sick leave benefits shall not be affected until required substantiation has been supplied.

(5) Medical and Dental Appointments

If a Part-time Employee requires time off for the purpose of attending a dental, physiotherapy,optical or medical appointment. provided **she** has been given prior authorization by the Employer. such absence shall be charged against her accumulated sick leave credits. Employees may be required to submit satisfactory proof of such appointment.

(6) **Request for Sick Credit** Balance

Upon request of an Employee, but not more frequently than once a year, the Employer shall advise an Employee of her accrued sick leave credits.

(7) Status When Sick Bank Exhausted

- (a) An Employee who has exhausted her sick leave credits during the course of an illness, and the illness continues, shall be deemed to be on leave of absence without pay or benefits except as provided in Article 27.01(c), for the duration of the illness or as provided below. The Employee shall keep the Employer advised as to when she may be expected back to work and shall provide the Employer with two (2) weeks written notice of her readiness to return to work. The Employershall then reinstate the Employee in the same position held by her immediately prior to the disability. At the expiration of twenty-four (24) months from the, last day of paid sick leave, an Employee who is not capable of returning to work pursuant to this Article shall be considered to have terminated her employement with the Employer.
- (b) The reinstatement of an Employee in accordance with this Article shall not be construed as being a violation of the posting and/or scheduling provisions of Articles 11 and 29.03,

ARTICLE30: TEMPORARYEMPLOYEES

- 30.01 (a) A Temporary Employeeshall be covered by the terms of this agreement with the exception of:
 - (i) Article 12 Annual Performance Appraisal
 - (ii) Article 32 -Layoff and Recall
 - (b) A Temporary Employee shall not have the right to grieve the termination of her employment on the expiry of the position for which she was hired or in accordance with other provisions of the Collective Agreement.
 - (c) The Employer shall provide at least seven (7) calendar days' written notice of termination of her term **position**, if she is **no** longer required in such position.

ARTICLE 31: CASUAL EMPLOYEES

31.01 Application of Collective Agreement

The provisions of this Collective Agreement shall not apply to Casual Employees except as provided by this Article.

31.02 Orientation

The Employer shall provide a paid orientation period for all new Casual Employees.

31.03 Hours of Work

- (1) The normal hours of work for a Casual Employee shall be up to seven and threequarter (7 3/4) hours in a day.
- (2) Casual Employees will not be required to work in excess of six (6) consecutive shifts except by routual agreement.
- (3) A Casual Employee will not be required to work in a manner where the ratio of work days to non-work days exceeds 5:2 averaged over six calendar weeks.
- (4) Hours of work shall be deemed to:
 - (a) include, as scheduled by the Employer, one (1) paid and one (I) unpaid rest periods of fifteen (15) minutes during each full working shift of seven and three quarter (7 3/4) hours; or
 - (b) one (I) rest period of thirty (30) minutes where fifteen (15) minutes are paid and fifteen (15) minutes are unpaid during each full working shift of seven and threequarter (7 3/4) hours if this is more compatible with scheduling of work assignments, the alternate to be applied shall be at the discretion of the Employes or
 - (c) include, as scheduled by the Employer, one (1) unpaid rest period of fifteen (15) minutes during each half shift of not less than four (4) hours; and
 - (d) exclude a meal period of thirty (30) minutes to be scheduled by the Employer during each working day on which the Employee works ir excess of four (4) hours.
 - (e) if an Employee is unable to take her meal period or rest period or is recalled to duty during her meal period or rest period she shall be given a full meal period or rest period later in her shift, or, where that is not possible, be paid for a meal period or rest period as follows:

• 39 •

- (i) for the rest period, one and one-half limes (11/2X) her Basic Rate of Pay rathet than at straight time; or
- (ii) for a meal period, at one and one-half times (1 1/2X) her Basic Rate of Pay.
- (5) No Casual Employee who works on a call-in basis shall be scheduled without her consent.
- (6) In the event that a Casual Employee reports to work as scheduled or called and the Employer cancelsher shift, the Employee shall be paid three (3) hours pay at her Basic Rate of Pay.

31.04 Overtime

- (a) Overtime is all time authorized by the Employer and worked by the Employee in excess of seven and three-quarter (7 3/4) hours per day and shall be paid at one and one-half times(1 1/2X) or two times (2X) her Basic Rate of Pay.
- (b) The overtime rate of one and one-half times (1 1/2X) times the Basic Rate of Pay shall be paid for work performed;
 - during the first four (4) hours worked in excess of seven and threequarters (73/4) in a day.
- (c) The overtime rate of two times (2X) the Basic Rate of Fay shall be paid for work performed
 - (i) after the first four (4) hours worked in excess of seven and threequarter (7 3/4) hours in a day.
 - (ii) in excess of seven and threequarter(73/4) hours worked on a named holiday.

31.05 Salaries

- (a) The basic rates **of** pay for Casual Employees shall be **as** outlined in the **Salaries** Schedule.
- (b) Casual Employees shall be entitled to the next level as provided in the Salary Schedule, upon the completion of two thousand, twenty-two and three-quarter (2022 3/4) hours worked

31.06

Shift Differential and Weekend Premium

- 40 -

- (a) A shift differential of seventy live cents (\$0.75) per hour shall be paid lo an Employee working on a shift Where the majority of hours of such shift fall within the period of fifteen hundred (1500) hours and zero seven hundred (0700) hours. Shift differential payment shall not be considered as part of the Employee's Basic Rate of Pay.

31.07

- all travel from the Facility t their 1 Casual Employees who of (a) n of their residen by means of public a following the shift, b who are r d fr gsoy ng] t remain C duty ic a than the regular shift and past the time when normal 1 transportation is 1 hall be t for the t of reasonable, expense from the Facility to their n :es and substantiated place of residence.
- (b) An up who is required t accompany a resident to another facility for medical reasons, shall be reimbursed for the cost of a and i j tr

31.08 Named Holidays

Ľ٤

э

Casual Employees all t pa at η and one-half times (1 1/2X) or : hours rk n the 1 toliday. A Casual Employee if all 1 paid in little to her Basic Rate of Pay four point two (4.2%) percent of such rate per pay period in lieu of e named h

31.09 Annual Vacations

(a) A Casual Employee shall be ith in addition to h Basic Rate of , four F % of her Basic R t of Fay in lieu of vacation, and shall be entitle to an additional two percent (2%) vacation pay upon the completion of the it hours of i i b it ime Employees to reach the vacation entitlement of fifteen (15) working days and a further two percent (2%) f i yon the juplet of equivalent hours of work required by a Full-time Employee to reach the vacation of two percent (2%) f or p y c th ti of equivalent hours f wc p ed by a Full-time Employee to reach the vacation of twenty five (25) working days

ľ

- (b) A Regular Employee shall not have her Level of Vacation Entitlement (i.e. 4%, 6%, etc.) reduced due to a change in status to a Casual Employee provided her service is uninterrupted.
- 31.10 Dues Deductions

Casual Employees shall be subject to dues deductions as provided in Article 4,

31.11 Grievance Procedure

Casual Employees shall be covered by the Grievance Procedure and Arbitration provisions of this Collective Agreement.

31.12 Appointments and Transfers

Subject to the criteria established in Article 11 of this Collective Agreement, an applicant for employment who has experience (accumulated hours) with the Employer as a Casual Employee shall be given preference over external applicants.

31.13 Performance Appraisal

Casual Employees defined in Article **2.05(b)(i), (ii)** and (iii) shall be covered by the Performance Appraisal Article of this Collective Agreement.

31.14 Disciplineand Dismissal

Casual Employees defined in Article **2.05(b)(i)**, (ii) and (iii) shall be covered by the Discipline and Dismissal Article of this Collective Agreement.

31.15 Shifts and Schedules

Employees shall be aware that in the course of their regular duties they may be required to work on various shifts throughout the twenty-four (24) hour period of the day and the seven (7) days of the week.

ARTICLE 32:LAYOFF AND RECALL

32.01 Joint Discussions

The Employer and the Union recognize the value of joint discussionswhen a layoff will occur. Representatives of the Employer and the Union may meet to discuss alternative layoff processes that may be more appropriate in the particular circumstances. In the event that the parties do not mutually agree in writing that alternative processes are appropriate, the fallowing will apply.

- 32.02 Layoff Process
 - (a) In reducing the workforce, Employees will be laid off in reverse order of seniority, within their classification subject to the following:

- (i) The remaining Employees have the ability to perform the work involved.
- (ii) An Employee cannot achieve a position in a higher paid classification through the operation of the lay-off provisions.
- (iii) A more senior Employee may be permitted to refuse a re-assignment and be laid off.
- (b) Temporary Employees shallbe released prior to Regular Employees being laid off, provided the Regular Employees have the ability to perform the work involved.
- 32.03 Notice
 - (a) Should a reduction in *the* working force occur, the Employer will notify Employees who **arc** to be laid off, at least fourteen (14) calendar days prior to the layoff, or shall grant pay in lieu thereof.
 - (b) The Union shall receive a copy of the notice of layoff forthwith.
- 32.04 Recall

When Employees are on layoff, the following process for re-call shall occur:

- Regular and temporary positions shall be posted and filled pursuant to Article 11. Employees on lay-off may apply for any posted vacancies.
- (b) When there are no applications from current RegularEmployees, the most seniorEmployee on layoff with the ability to perform the work involved shall be the first such Employee to be recalled.
- (c) No new Regular or Temporary Employees will be hired while there are other Employees on layoff who card be recalled in accordance with Article 32.04(b) above and awaiting recall.
- (d) The method of recall shall be by telephone and, if contact with the Employee is not accomplished, a letter/shall be sent by courier to the Employee's last known place of residence. The Employee so notified will return b work as soon as possible, but not later than five (5) days following the date of the telephone contact, or the date of the delivery of the letter.
- (e) A Regular Employee shall be considered terminated when she does not return from layoff as required or has been on layoff for a period of twenty four (24) months without being recalled
- (f) It is recognized and understood that it is the responsibility of the Employee to update the Employer about her whereabouts and to be available for recall.

- 43 -

32.05 Opportunities for Casual Work

- (a) The Employer shall endeavour to offer opportunities for casual work to laid off Employees in order of their seniority before assigning the work to a Casual Employee, providing the Employee has the ability io perform the work involved
- (b) Laid off Employees shall advise the Employer of their interest in and availability for such casual work at the time of layoff.
- (c) A laid off Employee may refuse an offer of casual work without adversely affecting her recall status.
- (d) A laid off Employee who accepts an offer of casual work shall be governed by the Collective Agreement provisions applicable to a Casual Employee, however, such Employee's recall status and ability to prepay benefits of certain contributorybenefit plans, and senioritystanding shall not be affected by the period of casual employment.
- 32.06 Benefit Coverage During Layoff

Employees affected by a layoff may elect to maintain coverage of contributoryplans specified in Article 22, subject to the insurers eligibility and other requirements, and provided **the** Employee makes prior arrangements to pay the full premium costs.

32.07 Application of Collective Agreement

The operation of this Article shall not be construed as a violation of the posting and/or scheduling provisions of the CollectiveAgreement.

32.08 Rights on Layoff

Other than for the pre-payment of certain contributory benefit premiums and the continuation of seniority held at the time of layoff, the Employee's rights on layoff shall be limited to the right of recall.

ARTICLE 33: DISCIPLINEAND DISMISSAL

- **33.01** Unsatisfactory conduct and/or performance by an Employee may be grounds for discipline up **to**, and including, immediate dismissal,
- **33.02** Unsatisfactory conduct and/or performance by an Employee which is not considered by the Employer to be **serious** enough to **warrant** suspension or dismissal may result in a written warning to the Employee. A copy of the written warning shall be placed on the Employee's personnel file. Copies of all written warnings shall be forwarded to the Unit President of the Union.

33.03	The Employee shall sign a written notice of discipline. for the sole purpose of indicating that she is aware of the disciplinary notice. An Employee may be accompanied by a representative of the Union during the disciplinary discussion.
33.04	When an Employee has grieved a disciplinary action and a designated Officer of the Employer has either allowed the grievance or reduced the penalty levied against the grievor, the personnel file of the Employee shall be amended to reflect this action provided that this action results in the abandonment of the grievance.
33.05	An Employee absent, for two (2) consecutively scheduled work days, without good and proper reason and without notifying the Employer, shall be considered to have terminated her services with the Employer.

Nothing in this Article prevents immediate suspension or dismissal for just cause. 33.06

ARTICLE 34: RESIGNATIONAND TERMINATION

34.01 Employee Notice of Termination

An Employee shall give the Employer at least twenty-eight (28) calendar days notice **d** termination of employment. 3

- 34.02 Vacation Pay Upon Termination .
 - If employment is terminated by the Employee without giving proper notice, pursuant IO Article 34.01 above, notwithstanding any other provisions of this Collective Agreement, such Employee shall receive vacation pay at the rate prescribed in the Employment Standards Code concerning vacation pay. The (a) Employer may waive this **Eleuse** if termination is due to illness of for any other reasons which are acceptable Io the Employer.
 - If employment is terminated, and proper notice given, the Employee shall (b) receive vacation pay in lieu of the unused period of vacation entitlement up to March 31 in each calendar year at the Employee's regular rate, together with the applicable percentage rate of vacation pay accrued on the Employee's regular earnings from the first day of April in the calendar year to the date of termination.
 - When an Employee is discharged for cause, vacation pay shall be at the rate (c) prescribed in the Employment Standards Code.

ARTICLE 35: HEALTH AND SAFETY

The Employer shall continue the Health and Safety Committee which shall be 35.01 composed of representatives of the Employer and two Unit representatives of the Union and may include representatives of other Employee groups. This committee shall meet once a month. An Employee shall be paid her Basic Rate of Pay for attendance at these committee meetings.

35.02 The Health and SafetyCommittee shall consider such matters as occupational health and safely and may make recommendations to the Employer in that regard. The Committee will function in accordance with the regulations published pursuant lo the Occupational Health and Safety Act or such other procedural rules as may be mutually agreed.

- 35.03 The Health and Safety Committee shall also consider measures necessary to ensure the security of each Employee on the Employer's premises and may make recommendations to the Employer in that regard. Should the tecommendations not be implemented and adequate steps taken towards implementation within two (2) months from the date the recommendation is made, the Health and Safety Committee may request and shall have the right to present the recommendation in writing, to the Health and Safety Committee within thirty (30)days of the presentation by the Committee.
- 35.04 The Occupational Health and Safety Committee will make recommendations to the Employer regarding possible strategies and processes in order to accommodate and promote early return to work of Employees who are recovering from illness or injury. In the event that the Union has recommendations that are different than the Occupational Health and Safety Committee, the Union shall have the right to submit such recommendations to the Employer.

ARTICLE 36: GRIEVANCE PROCEDURE

36.01 Grievance Definitions

A grievance shall be defined **as** any difference arising out of an interpretation. application, administration or alleged violation of this Collective Agreement. A grievance shall be categorized **as** follows:

- (a) an individual grievance is a dispute affecting one Employee. Such grievance shall be initiated at Step I of the grievance procedure as outlined in Article 36.06; or
- (b) group grievance is a dispute affecting two or more Employees. Such grievance shall be initiated at Step 2 and processed from there in the same manner as an individual grievance as outlined in Article 36.06. A group grievance shall list all Employees affected by the grievance and the results of such grievance shall apply proportionately if applicable, to all Employees listed on the original grievance; or
- (c) a policy grievance is a dispute between the parties which, due to its nature., is not properly the subject of an individual or group grievance. Such grievance shall be initiated, in writing, to the Department Head or the Union, by a representative of the grieved party within ten (10) days of the date the aggrieved party first became aware of or reasonably should have become aware of the event Lending to the grievance.

- 46

36.02 Authorized Representatives

- (a) An Employe8 may be assisted or represented by the Union or Unit Representative when presenting an official grievance.
- (b) The Employer agrees that the Unit Representative shall not be hindered, coerced or interfered with in any way in the performance of her function while investigating disputes and presenting adjustments as provided in this Article. However, no representative shall leave her work without obtaining consent from her supervisor, which shall not be unreasonably withheld The Unit Representative shall not suffer any loss of pay for the time spent in the performance of her duties involving a grievance provided that the representativedoes not leave the Employer's premises.

36.03 Time Limits

For the purposes of the Article, periods of time referred to in days shall be deemed such periods of time calculated on consecutive calendardays exclusive of Saturdays, Sundays and named holidays which are specified in Article 20. However, time limits may be extended by mutual agreement in writing.

36.04 Default

- (a) Should the Employee or the Union fail to comply with any time limits in the grievance procedure, the grievance will be considered **lo** be abandoned.
- (b) Should the Employer fail to comply with any time limits in the grievance procedure, the grievance shall automatically move to the next step on the day following the expiry of the particular time limit.

36.05 Mandatory Conditions

- (a) During any and all grievance proceedings. the Employee shall continue to perform her duties faithfully except in cases of suspension or dismissal.
- (b) A suspension or dismissal grievance shall commence at Step 2.

36.06 Steps in the Grievance Procedure

(a) Step 1

An Employee who has **a** complaint shall, within five (5) days of the date she becomes aware or reasonably should have become aware of **the** occurrence which led to **the** complaint. first discuss the matter with the Nursing Services Manager and attempt **to** resolve the complaint at this **stage**. In the event that it is not resolved, it may be advanced **as** an official grievance in accordance with the following steps.

- 47 -

(b) Step 2

Within five (5) days of discussing the complaint with the Nursing Services Manager, the Employee may submit an official grievance in writing stating the clause claimed to have been violated, the nature of the grievance, and the redress sought to the Director of Care Services who shall reply in writing within five (5) days of receiving the grievance. If the grievance is not settled at this stage, it may proceed to Step 3.

(c) Step 3

Within five (5) days of the reply from the Director of **Care** Services, the Employee may submit the grievance in writing to the Executive Director of the Facility or her designated representative. The Executive Director or his representative shall hold a hearing within five (5) days of receipt of the grievance. The Employee shall be entitled to have a representative of the **Unica** present during the hearing. The Executive Director or his representative shall render **a** written decision within five (5) days of the hearing. If the grievance is not settled at this stage, the Union may decide to proceed to Arbitration.

ARTICLE 37: ARBITRATION

37.01

(a) Either party wishing to submit a grievance to Arbitration shall, within ten (10) days of the receipt of the decision at Step 3 of the grievanceprocedure, notify the other party in writing of its intention to do so and name its appointee to the Arbitration Board, or state its desire to meet to consider the appointment of a single Arbitrator.

- (b) Within seven (7) days after receipt of notification provided for in Article 37.01(a) above, the party receiving notice shall:
 - (i) inform the other party of the name of its appointee to an Arbitration **Board;** or
 - (ii) arrange to meet with the other party in an effort to select a single Arbitrator. Where agreement cannot be reached on the principle and/or selection of a single Arbitrator. an Arbitration Board shall be established.
- (c) Where appointees to a Board have been named by the parties they shall, within seven (7) days, endeavourto select a mutually acceptable Chairman for the Arbitration Board. If they are unable to agree upon the choice of a Chairman, they shall immediately request the Director of Mediation Services to appoint a Chairman.

- (d) After a single Arbitrator has been selected, or the Arbitration Board has been formed in accordance with the above procedure. it shall meet with the patties as soon as possible tohear such evidence as the parties may desire topresent; assure a full. fair hearing and shall render the decision, in writing, to the parties at the earliest possible date following the hearing.
- (e) In the case of an Arbitration Board, the Chairman shall have the authority to render the decision with or without the concurrence of either of the other members and a decision thus rendered or the decision of the single Arbitrator shall be final and binding on all panics.
- (f) The Arbitration decision shall be governed by the terms of this Collective Agreement and shall not alter, amend or in any way rectify the terms of this Collective Agreement.
- (g) Each of the parties to this Collective Agreement shall bear the expenses of its appointee to the Arbitration Board. The fees and expenses of the Chairman or single Arbitrator shall be borne equally by the two parties to the dispute.
- (h) Any of the time limits herein contained in Arbitration proceedings may be extended if mutually agreed **to** in writing by the parties.

ARTICLE38: COPIES OF THE COLLECTIVE AGREEMENT

- 38.01 Within sixty (60) days of the signing of this Collective Agreement the Employer shall provide each Employee with a copy.
- 38.02 The Employer shall provide a copy of the Collective Agreement to each new Employee upon appointment.
- 38.03 The cost of producing such copies shall be shared equally between the Employer and the Union.

IN WITNESS WHEREOF the parties have executed this Collective Agreement by affixing hereto the signatures of their proper Officers in that behalf.

ON BEHALF OF YOUVILLE HOME

ON BEHALF OF UNITED STEELWORKERS OF AMERICA, LOCAL 5885

Alichard

M. Ternos

mayour Abou KB) ma

DATE: 12 Jan 200 /

DATE: Mail. 9/2007

- 50 -

SALARIES SCHEDULE

	<u>Step 1</u>	Step 2	Step 3	Step 4	Step 5	Step 6	Step 7	<u>Step 8</u>
Nursing Attendant								
April 1, 2000 August 1, 2000 April 1, 2001 August 1, 2001 April 1, 2002	10.16 10.16 10.57 10.57 10.89	10.83 10.83 11.26 11.26 11.60	11.21 11.21 11.66 11.66 12.01	11.62 11.62 12.08 12.08 12.44	12.03 12.03 12.51 12.51 12.89	12.45 12.45 12.95 12.95 13.34	12.85 13.36 13.36 13.76	13.76 14.17
Licensed Practical Nu	irse							
 April 1, 2000 August 1, 2000 April 1, 2001 August 1, 2001 April 1, 2002	12.95 13.47 14.01 14.57 15.01	13.50 14.04 14.60 15.19 15.64	14.05 14.61 15.20 15.80 16.27	14.60 15.18 15.79 16.42 16.90	15.15 15.76 16.39 17.04 17.53	15.67 16.30 16.95 17.63 18.16		

.....

- 51 -

LETTER OF UNDERSTANDING

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OF ST ALBERT

(hereinafter referred lo as the "Employer")

and

THE UNITED STEELWORKERS OF AMERICA LOCAL 5885 (hereinafter referred to as the "Union")

RE: OPTIONAL SCHEDULES

Whereas it is the desire of the membership of the Union to maximize the number of weekends that Employees **are** scheduled off, and

Whereas the Employer is willing to implementoptional schedules that attempt to accommodate the desire of the membership of the Union provided the employer's requirements regarding continuity of resident care, the viability of positions, and the needs of the organization, are met;

the Employer and the Union hereby agree as follows:

 (a) The provisions of Article 29.02(c)(ii) shall not apply to the positions currently occupied by Suheila Daw, Eva Wiebe, and Jana Hruza, and the following shall be substituted

"schedule **an** Employee not to work at least two weekends in a six (6) week period."

This amendment to Article **29.03(2)(c)(ii)**, applies to the positions and future incumbents of those positions.

- (b) Other pert time Employees may add their name to this list by notice in writing to the Employer.
- 2. Unless otherwise agreed by the Employer and the Employee, the full-time positions currently occupied by J. Lafreniere, A. Barradell, and L. Perrott, will be scheduled for one shift as a leave of absence without pay in a six week cycle, and in the event that such leave without pay occurs, it shall not be deemed to be a violation of the Collective Agreement nor be deemed to change the status of any incumbent Full-time Employee. This provision shall only apply as long as the above named individuals remain in the full-time positions they presently occupy.

- 52 -

3. Ongoing Scheduling Review and Evaluation

The parties acknowledge that it is desirable to ensure that the optional schedules are meeting the needs of both the Union and the Employer. Therefore the Employer and the Union shall develop an evaluation process in order to assess:

- (a) Employee satisfaction with the alternative shift schedules:
- (b) Impact on resident care; and
- (c) Organizational impact (i.e., turnover, orientation costs, etc.)
- 4. Business as Usual

The parties agree that this Letter of Understanding shall not be deemed to diminish any general rights of the Employer to manage the work force, as recognized in the Collective Agreement, and not withstanding **the** generality of the foregoing shall not be deemed to preclude the Employer from making decisions to adjust the working force from time to time.

5. TransitionalProvision

When alternateschedules are implemented or discontinued, the resultant changes shall not be deemed to be \mathbf{a} violation of the scheduling or posting provisions of the Collective Agreement.

ON BEHALF OF YOUVILLE HOME

ON BEHALF OF UNITED STEELWORKERS OF AMERICA, LOCAL 5885

timms

Jan 2.001

DATE:

DALA đ. iman

Niber 15-11 9/2001 DATE:

- 53 -

LETTER OF UNDERSTANDING

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OF ST. ALBERT

(hereinafter referred to as the "Employer")

and

THE UNITED STEELWORKERS OF AMERICA, LOCAL 5885 (hereinafter referred to as the "Union")

RE: BOOKING ADDITIONALSHIFTS AND HOURS FOR PART-TIME AUXILARY NURSING STAFF

WHEREAS both parties strive to deliver quality resident care and attempt to provide continuity of care with a minimum of service providers to the residents of Youville Home.

WHEREAS the parties are committed to respect the seniority of staff with the assignment of additional shifts where operationally possible, and would like to develop an efficient scheduling process with minimal administration to enhance these objectives.

Part-time Nursing Attendant Staff members on the unit where shifts/hours are available will be offered those available shifts/hours in order of seniority on the unit where the shifts/hours are available as follows:

- The parties agree to post in advance a three (3) month Part-time Availability Roster. Part-time Nursing Attendants will enter their availability on the posted Part-time Availability Roster beside their names, dates. types of additional shifts and hours available to work must be indicated Each month listed on the roster will be moved from the posting thirty (30) days prior to the occurrence of the month listed.
- 2. Youville Home will endeavour to contact the most senior Nursing Attendant staff in person or by telephone for available shifts. If no response is received to the telephone call or the staff member is not available the next senior part-time staff who has entered their availability as per Step I will be called, and **so** on down the list.
- 3. If attempts to contact part-time Nursing Attendants from the unit Availability Rosters are unsuccessful, Youville Home will proceed to the other unit's "Part-time Availability Roster" and commence calling Nursing Attendant staff in the same manner as indicated in Step 1 and Step 2 as per seniority on the other unit.

- 4. When replacing shift vacancies of more than four (4) shifts (and which may extend to up to three (3) months "long term absences"), Youville Home will replace the vacancy with a part-time staff member who is available to replace the entire period of that absence. Up to three (3)part-time staff will have the opportunity to cover absences which span fourteen (14) calendar days or more, with the submission of a group proposal. The proposal must commit the staff members to being scheduled for and work such additional shifts for the entire period of the absence. Part-time staff seeking additional shifts for long term absences of (14) calendar days prior to the commencement of the unforseen absence shall submit a request in writing within four (4) calendardays of such long term absence appearing on the schedule.
- 5. It is understood that the assignment of additional shifts among up to three (3) Part-time Employees to cover a long term absence will first be selected on the basis of total seniority in the p u p. In the event of a tie (more than one group with same total years of seniority) the group with the most senior person will be selected.
- The request for additional shifts and hours entered on the 'Part-timeAvailabilityRoster' and for proposals for filling long-term absences will not result in the payment of overtime.
- 7. When the above steps do not result in the provision of a schedule for additional shifts, and the Employer has not received a proposal for up to three (3) part-time staff to complete a long term absence which span fourteen (14) calendar days or more, the Youville Home will create a schedule with the most senice part-time staff member who has indicated she is available for part of the absence on the "Part-time Availability Rester", and if possible, will utilize up to two (2) other part-time Nursing Attendants who indicated they are available on the "Put-time Availability Rester" to schedule for the remainder of the absence, as long as continuity of care is not compromised in the opinion of the Employer.
- The Employer retains the option of contacting casual Nursing Attendants as per the casual protocol to complete schedules if it is not able to complete the schedule as per the above procedures.
- 9. The parties will meet as the need requires to discuss the process.

Either party may terminate this Letter of Understandingby providing the other party with sixty (60) days written notice of their intent to terminate the letter.

ON BEHALF OF YOUVILLE HOME

10.

ON BEHALF OF UNITED STEELWORKERS OF AMERICA, LOCAL 5885

M. Jimms

DATE: 12-001 2001

Frudy Milborty Chaine demas Mariane Debar B3/cu æ .

DATE _Poul &/2001_

- 56 -

LETTER OF UNDERSTANDING

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OF ST. ALBERT (hereinafter referred to as the "Employer")

and

THE UNITED STEEL WORKERS OF AMERICA LOCAL 5885 (hereinafter referred to as the "Union"

RE: COMMITTEETO DISCUSS ISSUES OF CONCERN

The parties agree that a meeting of Union and Employer representatives shall take place to discuss issues of concern relative to patient cate and other matters related to employment which are not covered by this Collective Agreement. Should the issues raised not be resolved, the Union may choose to make a written presentation of their recommendations to the B o d of Directors at an agreed upon time.

ON BEHALF OF YOUVILLE HOME

ON BEHALF OF UNITED STEELWORKERS OF AMERICA, LOCAL 5885

Inudy

Dicka m. Jimmo

6 prince Huma Thanjoni Deban 0 (IB

Dechaly

æ 19/2001 DATE: In

DATE: 12 Man 2001

- 57 -

LETTER OF UNDERSTANDING

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OFST. ALBERT (hereinafterreferred to as the "Employer")

and

THE UNITED STEEL WORKERS OF AMERICA LOCAL 5885 (hereinafter referred to as the "Union")

RE: HOURS OF WORK AND LAYOFFS

- Should it be necessary to reduce the hours of work or layoff during the life of this Collective Agreement. the senior person will be least affected by the reduction of hours or layoff.
- 2. If reductions in hours or layoffsare to occur; the Employer and the Union will meet within fourteen (14) calendar days of the announced reduction to discuss alternatives to the reduction in the hours of work or *the* layoffs.
- 3. If alternative measures cannot be agreed upon between the Employer and the Union within five (5) calendar days of the meeting telerred to in #2 above, the layoffs or reductions in hours will occur affecting least senior Employees prior to senior Employees.
- 4. Reductions to funding as a result of fluctuations in acuity (CMI) are not subject to the conditions of this Letter of Understanding.
- This Letter of Understanding is effective from the date of ratification until the date this Collective Agreement expires.

ON BEHALF OF YOUVILLE HOME

ON BEHALF OF UNITED STEELWORKERSOF AMERICA. LOCAL 5885

m. Jimms

12 Jan 2001 DATE:

ē,

DATE: Lepuil 9/2001

- 58 -

LETTER OF UNDERSTANDING

BETWEEN

THE YOUVILLE HOME (GREY NUNS) OF ST, ALBERT (hereinafter referred to as the "Employe")

and

THE UNITED STEELWORKERSOF AMERICA LOCAL \$885 (hereinafterreferred to as the "Union")

RE: ADMINISTRATION OF EDUCATIONAL ALLOWANCE FOR NURSING ATTENDANTS

The Parties agree as follows:

- This Letter of Understanding shall be applicable only to an Employee who is employed in the Nursing Attendant classification.
- For the purpose of determining an Employee's access to an educational allowance pay step. the Employer shall recognize the following courses or certificates, or equivalencies. as determined by the Employer, from bona fide post secondary institutions:
 - (a) Personal Care Attendant (PCA);
 - (b) Personal Support Aide (PSA);
- 3. An Employee shall be eligible to advance to Step 7 when:
 - (a) The Employee has successfully completed a recognized course or certificate, as outlined in Point 2 above; and
 - (b) The Employee has. in the case of a Full-time Employee, completed one (1) year of service at Step6, or in the case of a Part-time Employee, completed two thousand and twenty-two point seven five (2,022.75) regular hours actually worked at Step 6.
- 4. An Employee shall be eligible to advance to step 8 when:
 - (a) The Employee has successfully completed a recognized course or certificate, as outlined in Point 2 above; and
 - (b) The Employee has, in the case of a Full-time Employee, completed one (I) year of service at Step 7, or in the case of a Part-time Employee. completed two thousand

and twenty-two point seven five (2,022.75) regular hours actually worked at Step 7.

- 5. If a new Employee is hired as P nursing attendant as outlined in Point 1 above, on or after August 1, 2000, and the new Employee has successfully completed one (1) or more recognized courses or certificates as outlined in Point 2 above, such course work shall be recognized as one (I) yearly unit of previous experience.
- Employees who, as of August 1, 2000, are at Step 1 on the Salary Schedule, and have successfullycompleted a recognized course or certificate as outlined in Point 2 above, shall be eligible to advance to Step 2.
 - (a) RegularFull-timeEmployees who advance to Step 2 shall have their anniversary date, for the purpose of an annual increment, changed to August 1, 2000.
- 7. Employees who are not entitled to advance:
 - (a) to Step 7 in accordance with Point 3 of this Letter of Understanding; or
 - (b) to Step 8 in accordance with Point 4 of this Letter of Understanding; or
 - (c) to Step 2 in accordance with Point 6 of this Letter of Understanding

shall advance salary increments in accordance with the provisions of this Collective Agreement.

ON BEHALF OF YOUVILLE HOME

2 7n. mand

12 man 200

DATE:

LOCAL 5885 27. breit ma Deber I'n andere

ON BEHALF OF **UNITED** STEELWORKERS OF AMERICA,

9/2001 DATE:

LETTER OF UNDERSTANDING

BETWEEN

THE YOUVILLE HOME (GREYNUNS) OF ST. ALBERTA (hereinafterreferred to as the "Employer")

and

THE UNITED STEELWORKERSOF AMERICA LOCAL 5885

(hereinafter referred to as the "Union")

SALARY APPENDIX

The parties hereby agree to the following:

- All Full-time Employees who are currently at Step 4 and who have worked more than two thousand twenty-two point seven five (2,022.75) regular hours actually worked at Step 4 as of April 1, 2000 will be moved to Step 5 as of April 1, 2000. Regular Full-time Employees who advance to Step 5 shall have their anniversary date, for the purpose of annual increments, changed to April 1, 2000.
- All Part-time, Temporary or Casual Employees who are currently at Step 4 and who have worked more than two thousand and twenty-two point seven five (2,022.75) regular hours actually worked at Step 4 as of April 1, 2000 will be moved to Step 5 as of April 1, 2000.

ON BEHALF OF YOUVILLE HOME

1. 72 mg

DATE: 200 12

ON BEHALF OF UNITED STEELWORKERS OF AMERICA. LOCAL5885

l Wa

DATE: 2001